KANSAS BEGSTER

State of Kansas

BILL GRAVES Secretary of State

Vo	l. 6, No. 4	January 22, 1987	Pages 65-96
N	THIS ISSUE		Page
	Notice of Hearing on Proposed A Request for Proposals	dministrative Regulations	
	State Records Board Notice of Meeting		67
	Department of Health and Environm Public Notice Concerning Hazard	nent lous Waste Storage Permit	67
	State Board of Education Notice of Hearing on the Summe Notice of Hearing on Proposed A	r Food Service Program for Children	
	State Board of Accountancy Notice of Meeting	· · · · · · · · · · · · · · · · · · ·	68
	Attorney General Opinions No. 87-1 through 87-6		69
	State Grain Inspection Department Notice of Meeting	· · · · · · · · · · · · · · · · · · ·	
	Department of Administration Notice of Commencement of Neg	otiations for Architectural Services	70
	Department of Transportation Notice to Contractors		70
	Legislative Bills Introduced Through	January 14	
	State Board of Mortuary Arts Notice of Meeting		74
٠	State Corporation Commission Notice of Hearing	· · · · · · · · · · · · · · · · · · ·	
	Notice to Bidders for State Purchase	s	75
	Coffeyville Community College Dist Notice of Intent to Issue Revenue	trict e Bonds	76
	Notice of Bond Redemption City of Miltonvale City of LaHarpe		
	Notice of Bond Sale City of Hutchinson U.S.D. 200, Greeley County		
	lansing certain amounts appro	ropriations for the fiscal year ending June 30, 1987; priated from the state general fund for certain state in restrictions and limitations	84

State of Kansas SOCIAL AND REHABILITATION SERVICES

NOTICE OF MEETING

The Department of Social and Rehabilitation Services will meet at 9 a.m. Tuesday, February 3, in the Staff Development Training Center, 2700 W. 6th, Topeka.

The scheduled agenda includes:

—Announce discussions are underway for fiscal year 1989 Mental Health and Retardation Services budget.

—Discuss governor's budget.

—Present proposed State Economic Opportunity Office Federal Weatherization Plan.

The public is invited to this meeting. Telephone hook-ups are provided at the following locations of Social and Rehabilitation Services offices: Chanute, Emporia, Garden City, Hays, Hiawatha, Hutchinson, Junction City, Kansas City, Lawrence, Olathe, Osawatomie, Parsons, Pittsburg, Pratt, Salina, Topeka (area office and Docking State Office Building), Wichita and Winfield.

ROBERT C. HARDER Secretary of Social and, Rehabilitation Services

Doc. No. 004975

66

State of Kansas SOCIAL AND REHABILITATION SERVICES

NOTICE OF HEARING ON PROPOSED ADMINISTRATIVE REGULATIONS

The Department of Social and Rehabilitation Services will meet at 9 a.m. Friday, February 6, in the SRS board room, sixth floor, Docking State Office Building, Topeka.

The scheduled agenda includes:

—Public hearing concerning a proposed temporary administrative regulation. A summary of the proposed regulation is set forth below. The proposed change is scheduled to become effective upon approval by the State Rules and Regulations Board.

-Adoption of proposed temporary administrative

regulation.

1. 30-5-81b. The basis of reimbursement for hospital services. This regulation is being amended to reinstate the quarterly premium which had been deleted effective January 1, 1987.

The public is invited to the hearing.

ROBERT C. HARDER Secretary of Social and Rehabilitation Services

Doc. No. 004973

The Kansas Register is an official publication of the State of Kansas, published by authority of K.S.A. 75-430. The Kansas Register is published weekly by the Kansas Secretary of State, State Capitol, Topeka, KS 66612-1594. One-year subscriptions are \$47.50. Single copies may be purchased, if available, for \$2 each. Second class postage paid at Topeka, KS. ISSN No. 0744-2254.

Postmaster. Send change of address form to Kansas Register, Secretary of State, State Capitol, Topeka, KS 66612-1594.

© Kansas Secretary of State 1987. Reproduction of this publication in its entirety or for commercial purposes is prohibited without prior permission. Official enactments of the Kansas Legislature and proposed and adopted administrative regulations of state agencies may be reproduced in any form without permission.

PUBLISHED BY BILL GRAVES Secretary of State 2nd Floor, State Capitol Topeka, KS 66612-1594



Phone: (913) 296-3489

STATE HISTORICAL SOCIETY STATE RECORDS BOARD

NOTICE OF MEETING

The Kansas State Records Board will meet at 10 a.m. Thursday, January 29, in the conference room on the fourth floor of the Memorial Building, 120 W. 10th, Topeka. The board will consider requests from state agencies submitting proposals for disposition of noncurrent government records.

In addition, general administrative matters and

other business will be discussed.

EUGENE D. DECKER State Archivist Secretary, State Records Board

Doc. No. 004971

State of Kansas SOCIAL AND REHABILITATION SERVICES REQUEST FOR PROPOSALS

The Department of Social and Rehabilitation Services will accept applications for the provision of targeted assistance grant services to refugees for the grant period April 1, 1987 to March 31, 1988. The project area is limited to Sedgwick County. Applications must be received in the Wichita SRS area office by 5 p.m. Monday, February 2. Applications received after that date will not be considered.

Instructions for completing an application are described in an RFP packet available on request from John Sullivan, Department of Social and Rehabilitation Services, 3244 E. Douglas, P.O. Box 1620, Wichter 1700 (2012) 271 7000

ita 67201, (316) 651-5300.

ROBERT C. HARDER Secretary of Social and Rehabilitation Services

Doc. No. 004974

State of Kansas

DEPARTMENT OF HEALTH AND ENVIRONMENT

PUBLIC NOTICE

The U.S. Environmental Protection Agency (EPA), Region VII, 726 Minnesota Ave., Kansas City 66101, and the Kansas Department of Health and Environment (KDHE), Forbes Field, Topeka 66620, may issue a joint hazardous waste storage permit to the U.S. EPA Region VII Laboratory of Kansas City, Kansas.

The permit would allow for the operation of three storage areas for wastes generated from laboratory analytical activities, unneeded samples, and field

sampling activities.

The EPA and KDHE have jointly reviewed the hazardous waste permit application from the U.S. EPA Region VII Laboratory, 25 Funston Road, Kansas City, Kansas. If a decision is made to issue the joint permit,

the EPA portion of the permit would be issued under the authority of the Resource Conservation and Recovery Act (RCRA), and the KDHE portion of the permit would be issued under the authority of Kansas Statutes Annotated 65-3431(i) and 65-3431(s). Kansas has received final authorization to operate all portions of the RCRA permit program, except for the portions covered by the Hazardous and Solid Waste Amendments of 1984 (HSWA). Therefore, the EPA portion of the joint permit will address the requirements of HSWA, in particular Section 222. This section, effective on January 14, 1985, deals with the requirements and permit conditions involving the following wastes: F020, F021, F022, F023, F026, and F027.

Copies of the administrative record, which include the draft permit, permit application and all information submitted by the U.S. EPA Region VII Laboratory, are available for public review starting January 23 at the EPA Region VII Headquarters, 726 Minnesota Ave., Kansas City 66101, from 7:30 a.m. to 4:30 p.m. weekdays; and the KDHE, Room 131, Building 321, Forbes Field, Topeka, from 8 a.m. to 4:30 p.m.

weekdays.

Comments on the proposed permit may be directed in writing to Lyndell Harrington, RCRA Branch, EPA Region VII, Kansas City, Kansas, or to John Goetz, Hazardous Waste Section, KDHE, Topeka. Comments must be submitted in writing prior to March 8. Requests for additional information, including the fact sheet, may be made by contacting EPA at (913) 236-2888 or KDHE at (913) 862-9360.

A joint EPA and KDHE public hearing has been scheduled. It will be held at 7 p.m. Tuesday, March 10, in the second floor auditorium of the Kansas City-Wyandotte County Health Department, 619 Ann Ave.;

Kansas City, Kansas.

After consideration of all comments received and of the requirements of RCRA and state statutes, EPA and KDHE will make a final permit decision. If the decision is made to issue a joint permit that is substantially unchanged from those drafts made available for public comments as announced by this notice, the EPA regional administrator and the KDHE secretary will jointly notify all persons submitting comments or requesting notice of final decision. If the draft permit is substantially changed, the EPA regional administrator and the KDHE secretary will jointly issue a public notice indicating the revised decisions.

JACK D. WALKER, M.D. Secretary of Health and Environment

BOARD OF EDUCATION

NOTICE OF HEARING ON THE SUMMER FOOD SERVICE PROGRAM FOR CHILDREN

The Kansas State Board of Education will hold a public hearing at 1:30 p.m. Tuesday, February 10, in Room 121 of the Kansas State Education Building, 120 E. 10th, Topeka, to consider the proposed Kansas State Plan for the Summer Food Service Program for

Children for fiscal year 1987.

The proposed plan sets forth management and administrative policies and procedures to meet U.S. Department of Agriculture requirements for administration of the program. Chapter One presents 1986 program participation data and projections for 1987 participation; Chapter Two describes outreach activities and nondiscrimination policies and procedures; Chapter Three describes the training and monitoring activities to be conducted by the State Department of Education; Chapter Four identifies activities to be followed to maximize the use of on-site meal preparation and school food service facilities; and Chapter Five outlines the procedures for sponsor approval and disbursing funds to participating sponsors. Charts projecting total meals and federal funds also are included. Chapter Six describes audit and appeal procedures, and Chapter Seven presents the staff and budget for the State Department of Education's administration of the program.

A copy of the plan may be obtained by contacting the secretary of the State Board of Education, Kansas State Education Building, 120 E. 10th, Topeka 66612,

prior to the date of the hearing.

On the date of the hearing, all interested persons will be given a reasonable opportunity to present their views or arguments, either orally or in writing, in regard to the proposed plan. Individuals or organizations that cannot appear at the hearing may submit to the secretary of the board, at least five days before the hearing, their views concerning the proposed plan. All comments received will be considered by the board. The hearing shall be conducted in compliance with the public hearing procedures of the board.

KANSAS STATE BOARD OF EDUCATION
By: Bill Musick

Chairman Certified by:

Dr. Harold L. Blackburn Kansas Commissioner of Education

Doc. No. 004969

State of Kansas BOARD OF ACCOUNTANCY

NOTICE OF MEETING

The State Board of Accountancy will meet at 9 a.m. Tuesday, January 27, in Room 237, 503 Kansas Ave., Topeka. Persons interested in agenda items or in attending this meeting should contact the board office at (913) 296-2162.

GLENDA SHERMAN Board Secretary

Doc. No. 004966

State of Kansas

BOARD OF EDUCATION

NOTICE OF HEARING ON PROPOSED ADMINISTRATIVE REGULATIONS

The Kansas State Board of Education will hold a public hearing at 1:30 p.m. Tuesday, February 10, in Room 121 of the Kansas State Education Building, 120 E. 10th, Topeka, to consider proposed changes in Kansas Administrative Regulations 91-12-25, 91-12-52, 91-12-53, 91-12-54, 91-12-55 and 91-12-58. These regulations concern special education.

The proposed amendments to these regulations would increase the class size or caseload limitations, or both. Editorial changes also are proposed. The proposed changes are to become effective July 1,

1987

A copy of each proposed regulation is being mailed to chief school administrators in the state and may be reviewed in their offices by interested persons. Also, a copy of each of the proposed regulations and its fiscal impact statement may be obtained by contacting the secretary of the State Board of Education, Kansas State Education Building, 120 E. 10th, Topeka 66612, prior to the date of the hearing.

On the date of the hearing, all interested persons will be given a reasonable opportunity to present their views or arguments, either orally or in writing, in regard to the proposed regulations. Individuals or organizations that cannot appear at the hearing may submit written comments to the secretary of the board. All comments received will be considered by the board. The hearing shall be conducted in compliance with the public hearing procedures of the board.

KANSAS STATE BOARD OF EDUCATION

By: Bill Musick
Chairman
Certified by:
Dr. Harold L. Blackburn
Kansas Commissioner of Education

ATTORNEY GENERAL

Opinion No. 87-1

Intoxicating Liquors and Beverages—Cereal Malt Beverages—Off-Premise Sales. Edwin Wheeler, Jr., City Attorney, Marion, January 9, 1987.

Cereal malt beverage (CMB) licenses must state the place of business at which the beverages are to be sold. Delivery and retail sale of CMB to persons at their individual residences does not comply with the requirements of the licensing statutes. Cited herein: K.S.A. 41-2701; K.S.A. 41-2702; 84-2-105; 84-2-106; 84-2-401. TL

Opinion No. 87-2

Constitution of the State Kansas—Judicial—Compensation of Justices and Judges; Certain Limitation. Senator Robert G. Frey, 38th District, Liberal, January 9, 1987.

The term "salaried officer of the state," as used in Article 3, Section 13 of the Kansas Constitution, refers to any person holding a salaried state office. Common law criteria set out in *Durflinger v. Artiles*, 234 Kan. 502 (1983) must be applied to determine whether a person holding a specific state position is a state officer. Cited herein: K.S.A. 25-2505, 75-4301; L. 1972, ch. 390 and 392; Kan. Const., Art. 1, § 15; Kan. Const., Art. 3, § 13. TRH

Opinion No. 87-3

Consumer Credit Code—Insurance; Insurance in General; Consumer Credit Insurance; Property and Liability Insurance. Judith K. Stringer, Consumer Credit Commissioner, Topeka, January 9, 1987.

A creditor may not contract for or receive a separate charge for insurance unless the credit insurance provides a benefit which covers a loss to property related to the credit transaction. Household goods cannot be related to the transaction unless a purchase money credit arrangement is involved. Nor may the creditor be named as a loss payee of a policy which covers all of a consumer's household goods, even though that creditor has a security interest in some of the property covered under the policy. The amount of insurance is limited to the lesser of either the amount of debt or the value of the property insured. If a separate charge is received for credit insurance, before the insurance may be written, the consumer must be given an informed, affirmative, written request for the policy. In addition, a creditor may not sell credit insurance to a consumer if he has knowledge that an existing policy gives full coverage of the property related to the transaction, and the credit insurance would therefore be pro-rated. Cited herein: K.S.A. 16a-1-101, 16a-1-102; 16a-2-501; 16a-3-301; 16a-3-302; 16a-4-104; 16a-4-105; 16a-4-106; 16a-4-109; 16a-4-112; 16a-4-202; .6a-4-301; 16a-4-302; 50-626(b)(3); K.A.R. 40-5-8; 40-5-103; 15 U.S.C. §§ 45, 1605(b); 12 C.F.R. §§ 227.1; 227.13; 226.4; 226.18; 16 C.F.R. § 444.2(a)(4). JLM

Opinion No. 87-4

Laws, Journals and Public Information—Records Open to Public—Abstracts or Copies of Records; Fees. Gene Porter, Barton County Attorney, Great Bend, January 13, 1987.

The Kansas Open Records Act provides that public agencies may charge reasonable fees for photocopying public records. Fees are reasonable if they do not exceed the actual cost, or costs directly incurred, in providing copies. The twenty cents per page fee charged by Unified School District No. 431 is not unreasonable as it reflects the actual costs incurred in furnishing copies of public records. Cited herein: K.S.A. 45-215; 45-216; 45-219; 5 U.S.C. § 552. RLN

Opinion No. 87-5

Taxation—Property Exempt From Taxation—Claim to be Filed Each Year.

Constitution of the State of Kansas—Finance and Taxation—Exemption of Property for Economic Development Purposes. E. A. Mosher, Executive Director, Topeka, January 13, 1987.

Under the provisions of K.S.A. 1986 Supp. 79-213, the Board of Tax Appeals is authorized to examine the legal and factual basis of an economic development tax exemption granted pursuant to Article 11, Section 13 of the Kansas Constitution. However, the board has no authority to review the advisability of granting a proposed exemption, as that policy decision has been delegated to the governing body of the city or county by Article 11, Section 13 of the Kansas Constitution. Cited herein: K.S.A. 1986 Supp. 79-213; Kan. Const., Art. 11, § 13. TRH

Opinion No. 87-6

Crimes and Punishment—Code; Crimes Against the Public Morals—Sale of Magazines Containing Advertisements for Obscene Materials or Devices. Timothy Chambers, Reno County Attorney, Hutchinson, January 13, 1987.

A clerk selling an adult magazine can be held liable for promoting obscenity if the inclusion of an advertisement for obscene materials or devices makes the whole magazine obscene. Cited herein: K.S.A. 1986 Supp. 21-4301. BLB

ROBERT T. STEPHAN Attorney General

State of Kansas GRAIN INSPECTION DEPARTMENT

NOTICE OF MEETING

The Kansas State Grain Inspection Advisory Commission will meet at 9:30 a.m. Wednesday, January 28, in the Grain Inspection Department's conference room, eighth floor, Jayhawk Tower, 700 Jackson, Topeka.

Items on the agenda include a general discussion concerning the fiscal year 1987-88 budget and a possible fee increase for services provided to meet budget needs as required; and discussion of U.S.D.A.'s rule changes regarding additional net worth requirements on grain elevators under the Uniform Grain Storage Agreement and proposal of said increase to the legislature for uniformity.

GARY M. BOTHWELL Acting Director

Doc. No. 004986

State of Kansas

DEPARTMENT OF ADMINISTRATION DIVISION OF ARCHITECTURAL SERVICES

NOTICE OF COMMENCEMENT OF NEGOTIATIONS FOR ARCHITECTURAL SERVICES

Notice is hereby given of the commencement of negotiations for architectural services for minor building renovation for a PBX room at the Liberal Arts and Sciences Building at Wichita State University, Wichita.

Interested firms must have a Kansas office staffed by one or more architects licensed to practice in the state.

Any questions or expressions of interest should be directed to John B. Hipp, Division of Architectural Services, 625 Polk, Topeka 66603, (913) 233-9367, prior to February 6.

JOHN B. HIPP, AIA Director, Division of Architectural Services

Doc. No. 004970

State of Kansas

DEPARTMENT OF TRANSPORTATION

NOTICE TO CONTRACTORS

Notice is hereby given that sealed proposals for the construction of bridge work in the following Kansas counties will be received at the office of the Chief of Construction and Maintenance, K.D.O.T., Topeka, until 10 a.m. C.S.T., February 19, 1987, and then publicly opened:

DISTRICT ONE—Northeast

Atchison/Doniphan—7-106 K-3080-01—K-7, north city limits of Atchison north to the junction of K-20, 11.2 miles, overlay. (State Funds)

Atchison/Jackson/Nemaha—9-106 K-3081-01—K-9, east city limits of Wetmore east to the west junction of U.S. 159, 16.6 miles, overlay. (State Funds)

Brown—36-7 K-2714-01—U.S. 36, Nemaha-Brown county line east to the east junction of U.S. 36 and U.S. 75, 2.9 miles, pavement reconstruction. (State Funds)

Brown/Nemaha—106 K-3085-01—U.S. 36 in Brown County and K-236 in Nemaha County, 10.5 miles, overlay. (State Funds)

Jackson—75-43K-2918-01—U.S.75, junction of U.S. 75 and K-16 then north, 7.7 miles, slurry seal. (State Funds)

Jefferson—44 K-3082-01—K-4 and U.S. 59 in Jefferson County, 21.3 miles, recycling. (State Funds)

Johnson—435-46 K-2699-01—I-435, west of the I-435 and U.S. 69 interchange east to the Kansas-Missouri state line, 5.7 miles, signing. (Federal Funds)

Johnson—46 U-1035-01—Switzer Road bridge at Indian Creek in Overland Park, 0.1 mile, bridge replacement. (Federal Funds)

Marshall—58K-3079-01—U.S. 36, U.S. 77 and K-110 in Marshall County, 21.7 miles, overlay. (State Funds)

Marshall—36-58 M-1456-01—U.S. 36, Union Pacific Railroad east of Home City, east to K-87, 9.8 miles, patching. (State Funds)

Nemaha—36-66 K-2085-01—U.S. 36, North Fork Black Vermillion bridge 1, 1.1 miles east of the Marshall-Nemaha county line, bridge repair. (Federal Funds)

Nemaha—36-66 M-1468-01—U.S. 36, produce and stockpile hot mix material for KDOT mixing strip north of Seneca, stockpile material. (State Funds)

Riley—81 K-3083-01—U.S. 24 and K-113 in Riley County, 17.2 miles, overlay. (State Funds)

Riley—81 U-0831-03—Southern Arterial, K-18 east to U.S. 24 in Manhattan, 0.5 mile, grading and surfacing. (State Funds)

Wabaunsee—30-99 K-1806-01—K-30, from the I-70 junction north to Maple Hill, 2.0 miles, overlay. (State Funds)

Wyandotte—35-105 K-2909-01—I-35, 0.3 mile south of Johnson-Wyandotte county line northeast to K-12, lighting. (Federal Funds)

Wyandotte—105 U-0904-01—86th and Kansas Avenue in Kansas City, 0.3 mile, bridge replacement. (Federal Funds)

DISTRICT TWO—Northcentral

Clay—14 K-3066-01—K-15 and U.S. 24 in Clay County, 17.3 miles, overlay. (State Funds)

Dickinson/Marion—106 K-3068-01—U.S. 56 and U.S. 56B in Dickinson and Marion counties, 13.7 miles, overlay. (State Funds)

Ellsworth—70-27 M-1449-01—I-70, Safety Rest Area in the eastbound lane 2.5 miles east of the east junction of K-14, lagoon system modification. (State Funds)

Geary—31 K-3069-01—K-57, K-244 and K-244S in Geary County, 10.2 miles, overlay. (State Funds)

Morris—57-64 K-2894-01—K-57, Main Street north to the north city limits of Dwight, 0.4 mile, grading and surfacing. (State Funds)

Republic-79 U-0994-01-M Street, 14th to 15th

Street in Belleville, 0.1 mile, grading and bridge. (Federal Funds)

Washington—15E-101 M-1459-01—K-15E, 3.4 miles north of the junction of K-15E and K-243, 0.1 mile, culvert. (State Funds)

DISTRICT FOUR—Southeast

Chautauqua-10 K-3073-01-K-99 and U.S. 166 in Chautauqua County, 8.2 miles, overlay. (State Funds)

Chautauqua—166-10 K-0171-02—U.S. 166, Middle Caney Creek 10, 0.4 mile south of the south city limits of Sedan, 0.2 mile, bridge replacement. (Federal

Chautauqua—166-10 K-0171-04—U.S. 166, bridges 11, 12, 13, 14, 15 and 23, bridge replacement. (State Funds)

Chautauqua—166-10 K-2806-01—U.S. 166, Lake Creek bridge 16, 10.3 miles east of the east junction of K-99, bridge repair. (State Funds)

Chautauqua/Montgomery—166-106 K-3074-01—U.S. 66, west city limits of Niotaze east to the junction of U.S. 75, 4.6 miles, overlay. (State Funds)

Cherokee—11 K-3071-01—K-26, K-66, U.S. 69, U.S. 69A and U.S. 166 in Cherokee County, 15.1 miles, overlay. (State Funds)

Cherokee—26-11 K-2810-01—K-26, culvert 507, 3.1 miles north of U.S. 166, culvert. (State Funds)

Crawford—126-19 X-1034-02—K-126, Burlington Northern Railroad crossing at K-126 south of Beulah, grading and surfacing. (Federal Funds)

Labette-96-50 X-0988-02-K-96, Burlington Northern Railroad crossing at K-96 west of Altamont, grading and surfacing. (Federal Funds)

Labette-160-50 K-2898-01-U.S. 160, Leawood Street east to the east city limits of Parsons, 0.2 mile, grading and surfacing. (State Funds)

Labette/Montgomery-106 K-3072-01-U.S. 166 in Labette County and U.S. 166 and U.S. 169 in Montgomery County, 11.0 miles, overlay. (State Funds)

Montgomery-75-63 K-2818-01-U.S. 75, under Missouri Pacific Railroad bridge 13, 1.8 miles south of the Wilson-Montgomery county line, 0.2 mile, grading and surfacing. (State Funds)

Montgomery—166-63 K-2757-01—U.S. 166, 4.3 miles east of Tyro east to the west city limits of Coffeyville, 3.6 miles, recycling. (State Funds)

DISTRICT FIVE—Southcentral

Barber-4 K-3065-01-K-2, K-8 and U.S. 281 in Barber County, 23.4 miles, conventional seal. (State Funds)

Butler-177-8 K-2244-01-U.S. 177, Butler-Chase county line then south, 13.5 miles, conventional seal. (State Funds)

Cowley—15-18 K-2779-01—K-15, junction of U.S. 77 west to the east city limits of Udall, 5.9 miles, overlay. (State Funds)

Cowley-15-18 K-3064-01-K-15, Oklahoma-Kansas state line north to the east junction of U.S. 166 and U.S. 160 east to K-38, 10.8 miles, overlay. (State unds)

Cowley-55-18 X-0897-02-Crossing of Atchison, Topeka and Santa Fe Railroad and K-55 in Udall, grading and surfacing. (Federal Funds)

Cowley—15-18 M-1457-01—K-15, bridge approach for bridges 58, 59 and 61 west of the north junction of U.S. 77, surfacing. (State Funds)

Harper—39 K-3060-01—K-2 and U.S. 160 in Harper

County, 8.5 miles, recycling. (State Funds)

Harper/Kingman—14-106 K-3061-01—K-14, north junction of U.S. 160 north to the junction of K-42, 12.5 miles, conventional seal. (State Funds)

Kingman—14-48 K-2791-01—K-14, junction of K-42 north to 0.5 mile north of the south city limits of Kingman, 12.7 miles, conventional seal. (State Funds)

Sedgwick—54-87 K-2899-01—U.S. 54, intersection of U.S. 54 (Kellogg Avenue) and Maize Road in Wichita, 0.2 mile, intersection improvement. (State Funds)

Sedgwick-87 U-0965-02-MacArthur Road, Oatville to West Street in Wichita, 0.6 mile, grading and surfacing. (Federal Funds)

Sumner—96 K-3063-01—K-49 and U.S. 160 in Sumner County, 9.5 miles, surfacing. (State Funds) Sumner—55-96 K-2946-01—K-55, junction of U.S. 81 east to west city limits of Belle Plaine, 2.5 miles, overlay. (State Funds)

Proposals will be issued upon request to all prospective bidders who have been prequalified by the Kansas Department of Transportation on the basis of financial condition, available construction equipment, and experience. Also, a statement of unearned contracts (Form No. 284) must be filed. There will be no discrimination against anyone regardless of race, age, religion, color, sex, handicap or national origin in the award of contracts.

Each bidder shall file a sworn statement executed by or on behalf of the person, firm, association or corporation submitting the bid, certifying that such person, firm, association or corporation has not, either directly or indirectly, entered into any agreement, participated in any collusion, or otherwise taken any action in restraint of free competitive bidding in connection with the submitted bid. This sworn statement shall be in the form of an affidavit executed and sworn to by the bidder before a person who is authorized by the laws of the state to administer oaths. The required form of the affidavit will be provided by the state to each prospective bidder. Failure to submit the sworn statement as part of the bid-approval package will make the bid nonresponsive and not eligible for award consideration.

Plans and specifications for the projects may be examined at the office of the respective county clerk or at the K.D.O.T. district office responsible for the work.

> JOHN B. KEMP Secretary of Transportation

SOCIAL AND REHABILITATION SERVICES

PUBLIC ADVISORY COMMITTEE ON ENERGY ASSISTANCE AND CONSERVATION

NOTICE OF MEETING

The Public Advisory Committee on Energy Assistance and Conservation will meet at 9:30 a.m. Tuesday, February 3, at the Staff Development Building, 4th and Oakley, Topeka. The public is invited to attend and provide input into the developmental processes of the Low Income Energy Assistance Program and the Weatherization Assistance Program.

> ROBERT C. HARDER Secretary of Social and Rehabilitation Services

Doc. No. 004980

State of Kansas

LEGISLATURE

LEGISLATIVE BILLS INTRODUCED

The following lists the numbers and titles of bills and resolutions recently introduced in the Kansas Legislature.

Copies of bills and resolutions are available free of charge from the Legislative Document Room, 145-N State Capitol, Topeka 66612, (913) 296-4096. There is a limit of 25 copies of any one item.

Bills introduced through January 14:

HB 2001, by Representative Walker (by request): An act concerning elections; relating to precinct committeewomen and committeemen; amending K.S.A. 25-213 and repealing

HB 2001, by Representative Walker (by request): An act concerning elections; relating to precinct committeewomen and committeemen; amending K.S.A. 25-213 and repealing the existing section.

HB 2002, by Representative Rolfs: An act relating to property taxation; concerning certain hospital property exempt therefrom; amending K.S.A. 1986 Supp. 79-201b and repealing the existing section.

HB 2003, by Special Committee on Communication, Computers and Technology, re proposal No. 1: An act relating to taxation; exempting public television and radio systems from sales tax; amending K.S.A. 1986 Supp. 79-3606 and repealing the existing section.

HB 2004, by Special Committee on Energy and Natural Resources, re proposal No. 7: An act establishing the Kansas coal commission.

HB 2005, by Special Committee on Federal and State Affairs, re proposal No. 16: An act concerning recreation commissions; relating to the establishment and operation thereof; amending K.S.A. 1986 Supp. 12-16,102 and 79-2001 and repealing the existing sections; also repealing K.S.A. 1986 Supp. 12-16,102 and 79-2001 and repealing the existing sections; also repealing K.S.A. 1986 Supp. 12-1916, 12-1917, 12-1919 and 12-1921 and K.S.A. 1986 Supp. 12-1907, 12-1904, 12-1904, 12-1907 and 12-1908.

HB 2006, by Special Committee on Judiciary, re proposal No. 20: An act amending and supplementing the Kansas code for care of children; amending K.S.A. 38-1502, 38-1524, 38-1527, 38-1528 and 38-1542 and repealing the existing sections.

HB 2007, by Special Committee on Judiciary, re proposal No. 20: An act concerning children; relating to runaways; defining certain acts or omissions as crimes and providing penalties therefor; amending K.S.A. 1986 Supp. 21-3612 and repealing the existing section.

HB 2008, by Special Committee on Judiciary, re proposal No. 20: An act relating to certain crimes involving children; defining and classifying the crime of furnishing alcoholic beverages to a minor for illicit purposes; amending K.S.A. 186-54153 and repealing the existing

holic beverages to a fillion for finite purpose, and the existing section.

HB 2009, by Special Committee on Judiciary, re proposal No. 20: An act relating to the authority of the attorney general in certain criminal cases.

HB 2010, by Special Committee on Judiciary, re proposal No. 20: An act concerning the crime of indecent liberties with a child; amending K.S.A. 1986 Supp. 21-3503 and

crime of indecent liberties with a child; amending K.S.A. 1996 Supp. 21-3903 and repealing the existing section.

HB 2011, by Special Committee on Judiciary, re proposal No. 20: An act concerning certain homes for children; relating to qualifications therefor; amending K.S.A. 65-516 and repealing the existing section.

HB 2012, by Special Committee on Judiciary, re proposal No. 20: An act concerning missing persons; providing for the establishment of a missing children information clearinghouse; amending K.S.A. 75-712b and K.S.A. 1986 Supp. 75-712c and repealing the existing sections.

existing sections.

HB 2013, by Legislative Educational Planning Committee, re proposal No. 23: An act establishing the Kansas career-work-study program.

HB 2014, by Special Committee on Public Health and Welfare, re proposal No. 24: An act creating a commission on access to services for the medically indigent and the homeless; providing for the powers, duties and functions thereof.

HB 2015, by Special Committee on Public Health and Welfare, re proposal No. 25: An act concerning vocational programs serving the mentally retarded; placing limitations upon the approval or licensing thereof; amending K.S.A. 19-4001 and K.S.A. 1986 Supp. 75-3307b and repealing the existing sections.

HB 2016, by Special Committee on Public Health and Welfare, re proposal No. 25: An act concerning limitation on tax levies by counties; relating to mill levies for mental retardation; amending K.S.A. 79-1947 and repealing the existing section.

HB 2017, by Special Committee on Public Health and Welfare, re proposal No. 25: An act concerning adult residential living facilities; providing for the licensure thereof; providing for the administration of the licensure program; declaring certain acts to be

unlawful and classifying the crime and the penalties therefor; amending K.S.A. 19-2106, 36-501, 39-923, 40-2,116, 65-411, 65-3501 and 65-4702 and K.S.A. 1986 Supp. 12-4901 and 75-3307b and repealing the existing sections.

HB 2018, by Special Committee on Public Health and Welfare, re proposal No. 25: An act concerning adult family homes; relating to administration of registration of such homes; transferring certain powers and duties to the secretary of health and environment; amending K.S.A. 39-1501 and repealing the existing section.

HB 2019, by Special Committee on Public Health and Welfare, re Proposal No. 25: An act concerning the Kansas community mental retardation facilities assistance act; amending K.S.A. 1986 Supp. 65-4411, 65-4413 and 65-4414 and repealing the existing sections.

HB 2020, by Special Committee on Financing of Regents Institutions, re proposal No. 28: An act concerning state educational institutions under the control and supervision of the state board of regents; relating to residence requirements for fee purposes; amending K.S.A. 1986 Supp. 60-1903 and repealing the existing sections.

HB 2021, by Special Committee on Tort Reform and Liability Insurance, re proposal No. 29: An act concerning civil procedure; providing for itemized verdicts in certain cases; amending K.S.A. 1986 Supp. 60-1903 and repealing the existing section.

HB 2022, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act amending the uniform arbitration act; providing for arbitration of certain claims in tort; amending K.S.A. 5-401 and repealing the existing section.

HB 2023, by Special Committee on Tort Reform and Liability Insurance, re proposal No. 29: An act concerning governmental entities and their employees; relating to liability for certain acts and omissions; amending K.S.A. 75-6102, 75-6104, 75-6105, 75-6108 and 75-6112 and K.S.A. 1986 Supp. 75-6111 and 75-6116 and repealing the existing section.

HB 2024, by Special Committee on Tort Reform and Liability Insurance, re pr

therefor.

HB 2028, by Special Committee on Ways and Means, re proposal No. 36: An act relating to certain state officers and employees with law enforcement powers; amending K.S.A. 66:1319 and 74-5602 and K.S.A. 1986 Supp. 8-1910 and repealing the existing sections.

HB 2029, by Special Committee on Ways and Means, re proposal No. 38: An act concerning the Kansas public employees retirement system; relating to certain employees of the department of corrections; amending K.S.A. 74-4914a and K.S.A. 1986 Supp. 74-4914e and repealing the existing sections.

HB 2030, by Special Committee on Ways and Means, re proposal No. 38: An act concerning the Kansas public employees retirement system and certain retirement systems thereunder; relating to death and disability benefits; amending K.S.A. 74-4927e and 74-4960 and K.S.A. 1986 Supp. 20-2809, 74-4927 and 74-4959 and repealing the existing sections.

HB 2031, by Special Committee on Ways and Means, re proposal No. 38: An act concerning the Kansas public employees retirement system; relating to retirement date; amending K.S.A. 74-4915 and K.S.A. 1986 Supp. 74-4914 and 74-4937 and repealing the

existing sections.

HB 2032, by Special Committee on Ways and Means, re proposal No. 41: An act concerning county extension programs; relating to budget and program approval and priorities; appointment of agents; programs and personnel jointly funded by two or more counties; amending K.S.A. 2-615 and 2-616 and K.S.A. 1986 Supp. 2-610 and repealing the

existing sections.

HB 2033, by Special Committee on Ways and Means, re proposal No. 41: An act concerning retirement benefits for officers and employees of the state board of regents and institutions thereunder; relating to county extension agents; amending K.S.A. 1986 Supp. 74-4925c.

HB 2034, by Special Committee on Energy and Natural Resources, re proposal No. 10: An act concerning water; relating to the financing of water meter installation; amending K.S.A. 1986 Supp. 2-1915 and repealing the existing section.

HB 2035, by Special Committee on Energy and Natural Resources, re proposal No. 10: An act concerning conservation districts; relating to conservation structures; amending K.S.A. 1986 Supp. 2-1915 and repealing the existing section.

HB 2036, by Special Committee on Energy and Natural Resources, re proposal No. 10: An act concerning water; authorizing the purchase of a water right on a cost-sharing basis; amending K.S.A. 2-1908 and 82a-707 and K.S.A. 1986 Supp. 2-1915 and repealing the existing sections.

existing sections.

HB 2037, by Special Committee on Energy and Natural Resources, re proposal No. 10: An act concerning water; relating to floodplain regulation; amending K.S.A. 12-734 and 48-928 and repealing the existing sections.

HB 2038, by Special Committee on Legislative Apportionment, re proposal No. 21: An act relating to population in the state of Kansas; providing for the enumeration of the residents of the state; amending K.S.A. 1986 Supp. 11-201 and repealing the existing receivers.

section.

HB 2039, by Special Committee on Legislative Apportionment, re proposal No. 21: An act relating to elections; concerning the establishment and changing of election precincts; and prescribing certain powers and duties of the state and county election officers in relation thereto.

HB 2040, by Representative O'Neal: An act concerning children; relating to traffic infractions committed thereby; amending K.S.A. 38-1502 and 38-1602 and repealing the existing sections.

infractions committed unertoy, amending K.S.A. 1281 and Smith: An act concerning municipalities; relating to annexation; amending K.S.A. 12-519, 12-520b and 12-521 and K.S.A. 1986 Supp. 12-520 and 12-520a and repealing the existing sections.

HB 2042, by Representative Hensley: An act concerning the school district equalization act; authorizing the deposit of miscellaneous revenues in the general fund of a school district; conditions; amending K.S.A. 1986 Supp. 72-7062 and repealing the existing

section.

HB 2043, by Special Committee on Federal and State Affairs, re proposal No. 14: An act establishing a state lottery; providing for the administration and regulation thereof; amending K.S.A. 21-4302, 41-308, 74-5602, 75-4706 and 79-32,109 and K.S.A. 1986 Supp. 75-3732 and 79-3606 and repealing the existing sections.

HB 2044, by Special Committee on Federal and State Affairs, re proposal No. 14: An act relating to horse and greyhound racing and permitting parimutuel wagering thereon; providing for regulation and taxation thereof; prohibiting certain acts and providing penalties therefor; amending K.S.A. 21-4302, 38-1502, 38-1602 and 74-5602 and repealing the existing sections.

penatures therefore; amending sections.

HB 2045, by Representative Hensley: An act concerning biweekly payroll periods for state officers and employees; relating to approval thereof; prescribing certain powers, duties and functions for the secretary of administration; amending K.S.A. 75-5501a and

duties and functions for the secretary of administration; amending K.S.A. 75-5501a and repealing the existing section.

HB 2046, by Representatives Blumenthal and Sader. An act concerning school district finance; relating to increases in budgets per pupil for certain purposes; amending K.S.A. 1986 Supp. 72-7055, and repealing the existing section.

HB 2047, by Representatives Blumenthal and Sader: An act concerning school district finance; relating to limitations on budgets per pupil of certain districts; amending K.S.A. 1986 Supp. 72-7055, and repealing the existing section.

HB 2048, by Representative Kline: An act concerning certain counties; relating to the expenditure of county funds for small purchases; affecting competitive bidding requirements; amending K.S.A. 19-260b and repealing the existing section.

Commence of the Commence of th

HB 2049, by Committee on Appropriations: An act concerning appropriations for the fiscal year ending June 30, 1987; lapsing certain amounts appropriated from the state general fund for certain state agencies; and imposing certain restrictions and limitations. HB 2050, by Representative Fry: An act relating to radioactive materials; prohibiting its emplacement in bedded salt formations located within Kansas.

HB 2051, by Representative Wisdom (by request): An act concerning recreation systems; relating to the use of recreation facilities; amending K.S.A. 12-1903 and repealing the existing section.

existing section.

HB 2052, by Representatives Buehler, Graeber, Harper, O'Neal and Shore: An act concerning medical malpractice liability actions; relating to pretrial screening panels; amending K.S.A. 1986 Supp. 65-4904 and repealing the existing section.

HB 2053, by Representative Reardon: An act concerning the school district equalization act; affecting the definition of preschool-aged exceptional children as included in the definition of pupil; amending K.S.A. 1986 Supp. 72-7033 and repealing the existing section.

section.

SB 1, by Senators Ehrlich, Arasmith, Francisco, Frey, Hayden, Norvell, Reilly, Walker and Yost: An act relating to severance tax; concerning exemptions therefrom for the severance and production of oil; amending K.S.A. 1986 Supp. 79-4217 and repealing the

severance and production of oil; amending A.S.A. 1999 capp.

SB 2, by Senator Allen: An act concerning state educational institutions under the control and supervision of the state board of regents; relating to residence requirements for fee purposes and providing for a certain waiver therefrom; amending K.S.A. 76-729 and 76-730, and repealing the existing sections.

SB 3, by Senator Allen: An act concerning the state board of agriculture; amending K.S.A. 74-502, 74-503 and 74-520 and repealing the existing sections.

SB 4, by Senators Karr and Johnston: An act concerning the employment security act; relating to definition of certain terms; amending K.S.A. 44-703 and repealing the existing section.

SB 5, by Senators Karr and Johnston: An act amending the Kansas withholding and declaration of estimated tax act; concerning the definition of employee; amending K.S.A.

SB 5, by Senators Karr and Johnston: An act amending the Kansas withholding and declaration of estimated tax act; concerning the definition of employee; amending K.S.A. 79:3295 and repealing the existing section.

SB 6, by Special Committee on the Court System, re Proposal No. 5: An act concerning the retirement system for judges; relating to membership for district magistrate judges; relating to retirement benefits; amending K.S.A. 1986 Supp. 20-341, 20-2601 and 20-2610 and repealing the existing sections.

SB 7, by Special Committee on the Court System, re Proposal No. 5: An act concerning district court judges; relating to qualifications thereof; amending K.S.A. 1986 Supp. 20-334 and repealing the existing section.

SB 8, by Special Committee on the Court System, re Proposal No. 5: An act concerning judges; relating to the retirement system for judges; relating to contributions and calculation of final average salary; amending K.S.A. 1986 Supp. 20-2603 and 20-2610 and repealing the existing sections.

SB 9, by Special Committee on the Court System, re Proposal No. 5: An act concerning the district court; providing for the creation of certain district judge positions; amending K.S.A. 1986 Supp. 4-204, 4-211 and 4-219 and repealing the existing sections.

SB 10, by Special Committee on Energy and Natural Resources, re Proposal No. 6: An act relating to electric public utilities; providing for compensation to retail electric suppliers when service territory annexed by a city; amending K.S.A. 1986 Supp. 66-1,174 and 66-1,174 and 66-1,174 and 66-1,176 and repealing the existing sections; sB 11 by Special Committee on Federal and State Affairs, re Proposal No. 11: An act

66-1.176a.

SB 11, by Special Committee on Federal and State Affairs, re Proposal No. 11: An act concerning cemetery and funeral merchandise; relating to the sale thereof; requiring audits of certain funds; amending K.S.A. 1986 Supp. 16-320 and repealing the existing

SB 12, by Special Committee on Judiciary, re Proposal No. 17: An act concerning medical assistance; relating to determination of eligibility; concerning recovery of medical assistance paid; amending K.S.A. 21-3605, 39-709 and 39-719a and repealing the existing

Sections.

SB 13, by Special Committee on Judiciary, re Proposal No. 17: An act concerning medical assistance; relating to determination of eligibility; concerning recovery of medical assistance paid; amending K.S.A. 21-3605, 39-709 and 39-719a and repealing the existing

assistance paid; amending K.S.A. 21-3605, 39-709 and 39-719a and repealing the existing sections.

SB 14, by Special Committee on Judiciary, re Proposal No. 18: An act concerning criminal procedure; relating to diversion of proceedings on a complaint; amending K.S.A. 12417 and K.S.A. 1986 Supp. 8-2115, 12-4416, 12-4418, 22-2909, 22-2910 and 22-2911 and repealing the existing sections.

SB 15, by Special Committee on Judiciary, re Proposal No. 19: An act amending the Kansas juvenile offenders code; relating to dispositional alternatives; amending K.S.A. 38-1663 and repealing the existing section.

SB 16, by Special Committee on Judiciary, re Proposal No. 42: An act concerning civil procedure; relating to exemptions from legal process; amending K.S.A. 60-2304 and repealing the existing section.

SB 17, by Legislative Educational Planning Committee, re Proposal No. 23: An act concerning community colleges; relating to the determination of general state aid; affecting the definition of full-time equivalent enrollment; amending K.S.A. 1986 Supp. 71-611, 71-618, and 71-619, and repealing the existing sections.

SB 18, by Special Committee on Public Health and Welfare, re Proposal No. 26: An act concerning human body parts; declaring certain acts relating thereto not to be a sale and limiting liability therefor; amending K.S.A. 65-3701 and repealing the existing section.

SB 19, by Special Committee on Public Health and Welfare, re Proposal No. 26: An act concerning coroners; authorizing the removal of body parts for anatomical gift purposes; amending K.S.A. 19-1033 and repealing the existing section.

SB 20, by Special Committee on Public Health and Welfare, re Proposal No. 26: An act concerning the uniform anatomical gift act; amending K.S.A. 65-3209 and repealing the existing section.

existing section.

SB 21, by Special Committee on Public Health and Welfare, re Proposal No. 26: An act concerning the human body; declaring certain acts relating to the buying or selling of a human body or part thereof to be unlawful and classifying the crime and the penalties for violations.

SB 22, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act concerning insurance; relating to notice prior to increasing premiums for certain

29: An act concerning insurance; relating to notice prior to increasing premiums for certain policies.

SB 23, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act concerning insurance; relating to recording and reporting of loss and expense experience; creating advisory committee to the commissioner; amending K.S.A. 40-937 and 40-1118 and repealing the existing sections.

SB 24, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act concerning insurance; relating to the making of rates; amending K.S.A. 40-1112 and repealing the existing section.

SB 25, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act concerning tort liability for certain acts and omissions of persons and entities practicing as certified public accountants.

SB 26, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act amending the Kansas general corporation code; amending K.S.A. 17-6002 and 17-6305 and repealing the existing sections.

SB 27, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act concerning civil procedure; limiting civil liability of directors and officers of certain nonprofit organizations.

SB 28, by Special Committee on Tort Reform and Liability Insurance, re Proposal No. 29: An act concerning civil procedure; limiting civil liability of volunteers of certain nonprofit organizations.

the state historical society; relating to the closing and disposition of the General Frederick Funston memorial park; repealing K.S.A. 76-2025 and 76-2026.

SB 31, by Special Committee on Ways and Means, re Proposal No. 35: An act concerning acquisition of historic property by the state for preservation; amending K.S.A. 75-2726 and repealing the existing section.

SB 32, by Special Committee on Ways and Means, re Proposal No. 39: An act concerning the state park and resources authority; relating to motor vehicle permits; amending K.S.A. 1986 Supp. 74-45098 and repealing the existing section.

SB 33, by Special Committee on Ways and Means, re Proposal No. 40: An act relating to the state board of healing arts; concerning the composition thereof; amending K.S.A. 1986 Supp. 65-2812, 65-2813 and 65-2822 and repealing the existing sections.

SB 34, by Special Committee on Ways and Means, re Proposal No. 40: An act concerning the state board of healing arts; concerning the composition thereof; amending K.S.A. 65-2004, 65-2814 and repealing the existing sections.

SB 35, by Special Committee on Ways and Means, re Proposal No. 40: An act concerning the state board of healing arts; concerning the regulation of the practice of certain branches of the healing arts and related health care specialties by the board; authorizing certain procedure; amending K.S.A. 39:1504, 40:1126, 65-118, 65-2002, 65-2003, 65-2004, 65-2005, 65-2005, 65-2004, 65-2005, 6

SB 46, by Senator Gordon: An act concerning elections; relating to the recognition of political parties; amending K.S.A. 25-302a and repealing the existing section. HCR 5001, by Special Committee on Energy and Natural Resources, re Proposal No. 8: A concurrent resolution encouraging the development of a countywide water-wastewater

HCR 5001, by Special Committee on Energy and Natural Resources, re Proposal No. 8:
A concurrent resolution encouraging the development of a countywide water-wastewater management plan by each county.

HCR 5002, by Special Committee on Judiciary, re Proposal No. 20: A concurrent resolution directing certain state agencies to develop and require completion of educational programs and training for persons in certain positions likely to involve dealing with child sexual abuse cases.

HCR 5003, by Special Committee on Legislative Apportionment, re Proposal No. 21: A proposition to amend section 1 of article 10 of the constitution of the state of Kansas, relating to reapportionment of senatorial and representative districts.

HCR 5004, by Representative Hensley: A concurrent resolution relating to a joint rule for the house of representatives and senate for the 1987-88 biennium concerning deadlines for consideration of bills adjusting the state pay plan.

HCR 5005, by Representative Roe: A concurrent resolution relating to the joint rules for the senate and house of representatives for the 1987-88 biennium.

HR 6001, by Representatives Knopp and Barkis: A resolution relating to the organization of the House of Representatives.

HR 6003, by Representatives Roe: A resolution relating to the rules of the House of Representatives for the 1987-88 biennium.

HR 6004, by Representative Roe: A resolution relating to the rules of the House of Representatives for the 1987-1988 biennium.

KR 6005, by Representative Roe: A resolution adopting permanent rules of the House of Representatives Braden, Heinemann, Knopp and Moomaw: A resolution in memory of McDill "Huck" Boyd.

HR 6005, by Representative Roe: A resolution adopting permanent rules of the House of Representatives for the 1987-1988 biennium.

SCR 1601, by Special Committee on Public Health and Welfare, re Proposal No. 26: A concurrent resolution concerning the scope of hospital and physician services covered under the Kansas medicaid/medikan program; requesting the modif

standing committees.

SR 1802, by Senators Talkington and Johnston: A resolution relating to assignment of seats of the Senate.

SR 1803, by Senators Talkington and Johnston: A resolution relating to the organization

SR 1804, by Senator Arasmith: A resolution in memory of McDill "Huck" Boyd.

State of Kansas BOARD OF MORTUARY ARTS

NOTICE OF MEETING

The Kansas State Board of Mortuary Arts will conduct a special meeting at 1 p.m. Friday, January 30, in the conference room of the board's new office, 900 Jackson, Suite 856, Topeka.

DOUGLAS "MACK" SMITH Executive Secretary

Doc. No. 004967

State of Kansas

STATE CORPORATION COMMISSION

NOTICE OF HEARING

The State Corporation Commission has issued an order which penalized Murray Rodgers for failure to renew his operator or contractor license as required by K.S.A. 55-155 and K.A.R. 82-3-120.

Pursuant to K.S.A. 55-164, the matter is set for hearing at 11 a.m. Tuesday, February 3, in the conference room of the Conservation Division, 200 Colorado Derby Building, 202 W. 1st, Wichita.

JAMES E. BROWNE Assistant General Counsel

Doc. No. 004977

State of Kansas STATE CORPORATION COMMISSION

NOTICE OF MOTOR CARRIER HEARINGS

Applications set for hearing are to be heard at 9:30 a.m. before the State Corporation Commission, Docking State Office Building, fourth floor, Topeka, unless otherwise noticed.

This list does not include cases previously assigned hearing dates for which parties of record have received notice.

Questions concerning applications for hearing dates should be addressed to the State Corporation Commission, 4th Floor, Docking State Office Building, Topeka 66612, (913) 296-3808 or 296-2110.

Your attention is invited to Kansas Administrative Regulations (K.A.R.) 82-1-228, "Rules of Practice and Procedure Before the Commission."

Application set for February 3, 1987

Application for Extension of Certificate of Convenience and Necessity:

Gary Kettler, dba) Docket No. 152,942 M Kettler Trucking) Route 2, Box 96) Ottawa, KS 66067) MC ID No. 124129

Applicant's Attorney: None

Livestock.

Between all points and places in Wyandotte, Johnson, Miami, Franklin, Linn, Osage, Anderson and Douglas counties, Kansas, on the one hand, and on the other, all points and places in the state of Kansas.

Application set for February 12, 1987

Application for Transfer of Certificate of Convenience and Necessity:

Grimes Transport, Inc.

) Docket No. 28,724 M

1426 N. Summitt Arkansas City, KS 67005

) MC ID No. 100213

TO:

Holton Transport, Inc. P.O. Box 230

Holton, KS 66436

Applicant's Attorney: Don Hults, P.O. Box 225, Lawrence National Bank Building, Lawrence, KS 66044

Petroleum and petroleum products in bulk, in tank truck loads,

From all Kansas refineries and pipeline terminals, on the one hand, to points and places in the counties of Gray, Meade, Ford, Clark, Edwards, Kiowa, Comanche, Pratt, Barber, Kingman, Reno, Harper, Harvey, Sedgwick, Sumner, Butler, Cowley, Greenwood, Lyon, Wabaunsee, Elk, Chautauqua, Montgomery, Wilson, Woodson, Coffey, Osage, Shawnee, Franklin, Anderson, Allen, Neosho and Labette, on the other.

Aircraft fuel,

From any points within the aforesaid listed counties to any military post, base or reservation within the state of Kansas.

Petroleum products,

From any leaking storage tank or facility in aforesaid listed counties to any points in the state of Kansas.

From any underground storage point in aforesaid listed counties to any storage point in the state of Kansas.

Anhydrous ammonia, in bulk, in tank vehicles,

From all Kansas refineries, pipeline terminals and storage points, on the one hand, and on the other, to all points and places in Kansas.

Road oils.

From all refineries and storage facilities in Butler county to all points and places in Kansas.

WILLIAM E. GREEN
Administrator
Transportation Division

CONTRACT PROPERTY.

สองสมัย สิธิว

DEPARTMENT OF ADMINISTRATION DIVISION OF PURCHASES

NOTICE TO BIDDERS

Sealed bids for the following items will be received by the Director of Purchases, Landon State Office Building, 900 Jackson, Room 102, Topeka, until 2 p.m. C.S.T. on the date indicated, and then will be publicly opened. Interested bidders may call (913) 296-2377 for additional information.

MONDAY, FEBRUARY 2, 1987

#27526

Statewide—AUTOMOTIVE FILTERS

#68046

Department of Transportation—HOT MIX, PLANT MIX BITUMINOUS MIXTURE, various locations #68047

University of Kansas Medical Center— HYPER/HYPOTHERMIA UNIT

#68048

Kansas Fish and Game Commission—READY MIX CONCRETE, Miami State Fishing Lake

#68051

Kansas Fish and Game Commission—AGGREGATE AND CONCRETE, Valley Falls #68052

Department of Transportation—MUD JACK, Chanute

#68053

Department of Transportation—PNEUMATIC ROLLERS, various locations
#68057

University of Kansas Medical Center—BONE SURGERY INSTRUMENTATION
#68058

University of Kansas—LAB CENTRIFUGE #68094

Kansas State University—GRAIN

#68097

Kansas State University—SOIL MOISTURE DETERMINATION

TUESDAY, FEBRUARY 3, 1987

#A-5631

Department of Transportation—REROOF AREA BUILDING, Marion

#27538

Kansas Correctional Industries—RENDERING MATERIALS—SERVICE

#68054

Department of Transportation—FORK LIFT, Norton #68055

Department of Transportation—AIR COMPRESSOR, various locations

#68056

Department of Transportation—CRUSHED LIMESTONE, Johnson County #68064

Department of Transportation—CONCRETE SAW, Hutchinson

#68065
University of Kansas—FURNISH AND INSTALL
NEW DOCK DOORS

#68067

Kansas State University—TRUCK

#68068

Department of Transportation—SEDANS, various locations

#68069

Department of Transportation—VEHICLES, various locations

WEDNESDAY, FEBRUARY 4, 1987

#27527

Statewide—LARGE PHOTO AND STAGE STUDIO LAMPS

#27528

University of Kansas—CONCRETE CURING COMPOUND, EXPANSION JOINT MATERIAL #68072

Kansas State University—SEED COUNTER, Hays #68073

University of Kansas—FUEL OIL #68074

Department of Revenue—PRINTER/POWER DIRECTORS/TAPE BACKUP/FILE SERVER #68080

Kansas Fish and Game Commission—BOAT RAMP AT HILLSDALE RESERVOIR, Miami County #68081

Department of Health and Environment—ANALYTICAL SERVICES

#68082

University of Kansas—FURNISH AND INSTALL SPRAY ON CEILING

THURSDAY, FEBRUARY 5, 1987

#27534

Kansas Correctional Industries—TOWELING #68085

Department of Transportation—FURNISH AND INSTALL UNIT HEATERS, various locations #68086

Kansas State University—CARPET #68087

Department of Social and Rehabilitation Services—ENVELOPES—DOUBLE WINDOW

FRIDAY, FEBRUARY 6, 1987

#68091

University of Kansas Medical Center—REFURBISHING AIRPLANE

#68092

Department of Transportation—DIAMOND CORE BITS

#68093

Kansas State University—PLOT THRESHER #68095

Department of Human Resources—CONTINUOUS MAILERS

MONDAY, FEBRUARY 9, 1987

#27530

Statewide—MICROCOMPUTER SOFTWARE #27532

Statewide—ENVELOPES—UNPRINTED

MONDAY, FEBRUARY 16, 1987

#27397-A

Statewide—STATEWIDE TRAVEL SERVICES

TUESDAY, FEBRUARY 17, 1987

#A-5692

Department of Human Resources—REPLACE AIR CONDITIONING AND HEATING SYSTEM, Dodge City

MONDAY, FEBRUARY 23, 1987

#68083

Kansas Commission on Veterans' Affairs—LEASE OF FARMLAND

#68102

Larned State Hospital—LEASE OF FARMLAND

FRIDAY, FEBRUARY 27, 1987

#27461

Emporia State University—PROPERTY INSURANCE

NICHOLAS B. ROACH Director of Purchases

Doc. No. 004965

State of Kansas SOCIAL AND REHABILITATION SERVICES

FAMILY AND CHILDREN TRUST FUND SOLICITATION

Application kits will be available for the programs listed below from the SRS area offices beginning January 26, 1987:

Program I—Continuation Grants. The programs currently receiving first, second, or third year trust fund grants are eligible to apply for second, third, or fourth year grants. All of these applicants will be subject to the trust fund guidelines being issued January 26, 1987. Each program will be notified of the maximum amount of funds available to them in 1987.

Program II—New Community Based Child Abuse and Neglect and Family Abuse Prevention Grants. Approximately five to 10 new programs will be funded in dollar amounts ranging from \$500-\$20,000.

Program III—New Community Based Child Sexual Abuse Prevention Grants. Six projects will be funded for up to a maximum amount of \$20,000 each for the first year to further the goal of reducing the sexual abuse of children in Kansas. Three of the projects will be funded beginning July 1, 1987, and the second three beginning July 1, 1989. This program is limited to providers proposing to serve counties with at least 2,000 children and no more than 7,500 children.

Applications for Program I, Program II and Program III must be submitted to the SRS area office no later than 5 p.m. March 13. Late applications will not be considered. Final award decisions will be made by the Children and Youth Advisory Committee. Successful applicants will be notified about June 15, 1987.

Pre-submission conferences will be held in Topeka on Friday, February 6; in Hutchinson on Tuesday, February 10; and in Chanute on Thursday, February 12

Questions regarding this notification should be directed to the SRS area office social service chiefs or to David O'Brien of SRS/Youth Services, (913)296-4649.

ROBERT C. BARNUM Commissioner, Youth Services

(Published in the KANSAS REGISTER, January 22, 1987.)

NOTICE OF INTENT TO ISSUE REVENUE BONDS COFFEYVILLE COMMUNITY COLLEGE DISTRICT MONTGOMERY COUNTY, KANSAS

You are hereby notified that the Board of Trustees of the Coffeyville Community College District, Montgomery County, Kansas, did on January 12, 1987 duly adopt a resolution declaring it necessary to and authorizing the enlarging of the existing dormitory system of the Coffeyville Community College District by building and equipping an additional dormitory building or buildings to be located on the campus of the Coffeyville Community College, Coffeyville, Kansas, which will house approximately 50 students, at a total estimated cost of \$390,000 under the authority of K.S.A. 76-6a12 to 76-6a25, inclusive, as amended and supplemented.

You are hereby further notified that said resolution of the Board of Trustees of the Coffeyville Community College District, Montgomery County, Kansas, declares it necessary to and authorizes the issuance and sale of dormitory system revenue bonds of the Coffeyville Community College District, Montgomery County, Kansas, in an amount not to exceed \$390,000, in order to pay the costs of building and equipping said dormitory building or buildings.

You are hereby further notified that unless an action to contest the legality of the proposed revenue bonds of the Coffeyville Community College District, Montgomery County, Kansas, shall be filed in a court of law within 30 days of the date of publication of this notice, the right to contest the legality of any revenue bonds issued in compliance with the aforesaid resolution and other proceedings duly and legally had and taken by said Board of Trustees prior to the date of publication of this notice, and the right to contest the validity of the provisions of such proceedings, shall cease to exist, and no court shall thereafter have the authority to inquire into such matters; and that after the expiration of said 30 days from the date of publication of this notice, no one shall have any right to commence an action contesting the validity of such revenue bonds or the provisions of such proceedings of the Board of Trustees and all such revenue bonds shall be conclusively presumed to be legal and no court shall thereafter have the authority to inquire into such matters.

Dated January 12, 1987.

BOARD OF TRUSTEES
COFFEYVILLE COMMUNITY COLLEGE
DISTRICT, MONTGOMERY COUNTY, KANSAS
By Dean Daniels, Chairman
Attest: Dan Kinney, Secretary

Doc. No. 004982

(Published in the KANSAS REGISTER, January 22, 1987.)

NOTICE OF REDEMPTION CITY OF MILTONVALE, KANSAS INDUSTRIAL REVENUE BONDS SERIES 1977 (TRI-RIVER CABLE, INC.)

Notice is hereby given that pursuant to Section 3 of Ordinance No. 422 of the city of Miltonvale, First National Bank and Trust Co. of Salina, as trustee and paying agent will redeem on March 1, 1987 for the city, \$42,000 principal amount of the Series 1977 industrial revenue bonds of the city of Miltonvale (Tri-River Cable, Inc.) at a price equal to the principal amount thereof plus accrued interest thereon to March 1, 1987, plus a premium of 1 percent.

All bonds are to be redeemed and are listed as

follows:

vs:	
Bond Number	Amount
39	1,000
40	1,000
41	1,000
42	1,000
43	1,000
44	1,000
45	1,000
46	1,000
47	1,000
48	1,000
49	1,000
50	1,000
51	1,000
52	1,000
53	1,000
54	1,000
55	1,000
56	1,000
57	1,000
5 8	1,000
59	1,000
60	1,000
61	1,000
62	1,000
63	1,000
64	1,000
65	1,000
66	1,000
67	1,000
68	1,000
69	1,000
7 0	1,000
71	1,000
72	1,000
73	1,000
74	1,000
75 70	1,000
76	1,000
77 7 2	1,000
78	1,000
79	1,000
80	1,000

On March 1, 1987, the bonds described above will be due and payable at the First National Bank and Trust Company, 101 N. Santa Fe, P.O. Box 800, Salina, KS 67402-0800.

Bonds should be surrendered with all unmatured interest coupons attached. After March 1, 1987, interest on the aforesaid bonds will cease to accrue.

FIRST NATIONAL BANK AND TRUST COMPANY OF SALINA Trustee and Paying Agent

Doc. No. 004984

(Published in the KANSAS REGISTER, January 22, 1987.)

NOTICE OF REDEMPTION OF CITY OF LAHARPE, KANSAS GENERAL OBLIGATION SEWER SYSTEM BONDS SERIES A, 1982 DATED MARCH 1, 1982

Pursuant to Section 1 of Ordinance No. 1619, as amended by Ordinance 1619A of the city of LaHarpe, Kansas, notice is hereby given that the city has determined to call its outstanding general obligation sewer system bonds, Series A, 1982, dated March 1, 1982, for redemption and payment on March 1, 1987. The bonds which will be called for redemption and payment on that date are described and identified as follows:

Bond Numbers (Inclusive)	Principal Maturing Amount	Interest Rate Per Annum	Original Maturity Date
6 to 9	\$20,000	9.75%	September 1, 1987
10 to 14	25,000	10.00%	September 1, 1988
15 to 19	25,000	10.25%	September 1, 1989
20 to 24	25,000	10.50%	September 1, 1990
25 to 29	25,000	10.75%	September 1, 1991
30 to 34	25,000	11.00%	September 1, 1992
35 to 39	25,000	11.25%	September 1, 1993
40 to 44	25,000	11.50%	September 1, 1994

On the redemption date there shall become due and payable on each of the above described bonds (upon presentation and surrender of each such bond and all appurtenant coupons thereto) the redemption price thereof which is equal to 100 percent of the principal amount of each bond, together with interest accrued to the said redemption date. Interest shall cease to accrue on the bonds so called for redemption and payment from and after March 1, 1987; all interest coupons maturing after March 1, 1987 shall become void.

The bonds that have been called for redemption as described above, together with the appurtenant coupons, must be presented for payment and surrendered to the Treasurer of the State of Kansas, Topeka, Kansas, paying agent for the bonds.

Dated November 12, 1986.

THE CITY OF LAHARPE, KANSAS
By Mignonette L. Kelley
City Clerk

ŧ,

48 U

(Published in the KANSAS REGISTER, January 22, 1987.)

NOTICE OF BOND SALE \$1,065,000 INTERNAL IMPROVEMENT BONDS SERIES 1987-A OF THE

CITY OF HUTCHINSON, KANSAS (general obligation bonds payable from unlimited ad valorem taxes)

Sealed Bids

Sealed bids will be received by the undersigned, city clerk of the city of Hutchinson, Kansas, on behalf of the Commission at City Hall, 125 E. Avenue B, P.O. Box 1567, Hutchinson, KS 67504-1567, until 10 a.m., C.S.T. Tuesday, January 27, 1987, for the purchase of \$1,065,000 principal amount of internal improvement bonds, Series 1987-A, of the city hereinafter described. All bids will be publicly opened and read at said time and place and will be acted upon by the governing body immediately thereafter.

Bond Details

The bonds will consist of fully registered bonds in the denomination of \$5,000 or any integral multiple thereof, dated February 1, 1987, and becoming due serially on September 1 in the years as follows:

Year	-9 x 1 + 8	Principal Amount
1988		30,000
1989	* 1	30,000
1990		30,000
1991		35,000
1992	**	35,000
1993		40,000
1994		40,000
1995		45,000
1996		45,000
1997		50,000
1998		55,000
1999		55,000
2000		60,000
2001	1 1 1 1 1 1 2 1 2 2 2 2 2 2 2 2 2 2 2 2	70,000
2002	4.7	70,000
2003		80,000
2004		85,000
2005		90,000
2006		60,000
2007		60,000

The bonds will bear interest at rates to be determined when the bonds are sold as hereinafter provided, which interest will be payable semiannually on March 1 and September 1 in each year, beginning on September 1, 1987.

Place of Payment and Bond Registration

The principal of and interest on the bonds will be payable in lawful money of the United States of America by check or draft of the Treasurer of the State of Kansas, Topeka, Kansas (the paying agent and bond registrar), to the registered owners thereof whose names are on the registration books of the bond registrar as of the 15th day of the month preceding each interest payment date. The bonds will be registered pursuant to a plan of registration approved by the city

and the Attorney General of the State of Kansas. The bonds may be registered as fully registered certificated bonds or uncertificated (book entry) bonds at the option of each registered owner.

The city will pay for the fees of the bond registrar for registration and transfer of the bonds and will also pay for printing a reasonable supply of registered bond blanks. Any additional costs or fees that might be incurred in the secondary market, other than fees of the bond registrar, will be the responsibility of the bondholders.

The type and denominations of the bonds and the names, addresses and social security or taxpayer identification numbers of the registered owners shall be submitted in writing by the successful bidder to the city and bond registrar at least two weeks prior to the closing date.

The initial reoffering price to the public by the original purchaser shall be furnished to the city at least one week prior to the closing date. A certificate setting forth such initial reoffering price to the public shall be furnished to the city by the original purchaser at closing.

Redemption of Bonds Prior to Maturity

Bonds maturing in the years 1988 to 1998, inclusive, shall become due without option of prior payment. At the option of the city, bonds maturing in the years 1999 to 2007, inclusive, may be called for redemption and payment prior to maturity in whole or in part in inverse order of maturity (selection of bonds within the same maturity to be by lot by the city in such equitable manner as it may determine) on September 1, 1998, or on any interest payment date thereafter at the redemption prices (expressed as percentages of the principal amount) plus accrued interest to the redemption date:

Redemption Dates	Redemption Price
September 1, 1998, and March 1, 1999	103%
September 1, 1999, and March 1, 2000	102%
September 1, 2000, and March 1, 2001	101%
September 1, 2001, and thereafter	100%

Whenever the city is to select the bonds for the purpose of redemption, it shall, in the case of bonds in denominations greater than \$5,000, if less than all of the bonds then outstanding are to be called for redemption, treat each \$5,000 of face value of each such fully registered bond as though it were a separate bond of the denomination of \$5,000.

If the city shall elect to call any bonds for redemption and payment prior to the maturity thereof, the city shall give written notice of its intention to call and pay said bonds on a specified date, the same being described by number and maturity, said notice to be mailed by United States registered or certified mail addressed to the registered owners of said bonds, to the State Treasurer of Kansas, and to the manager or managers of the underwriting account making the successful bid; each of said notices to be mailed not less than 30 days prior to the date fixed for redemption. If any bond be called for redemption and payment as aforesaid, all interest on such bond shall cease from and after the date for which such call is made,

provided funds are available for its payment at the price hereinbefore specified.

Conditions of Bids

Proposals will be received on the bonds bearing such rate or rates of interest as may be specified by the bidders, subject to the following conditions: The same rate shall apply to all bonds of the same maturity. Each interest rate specified shall be a multiple of 1/8 or 1/20 of 1 percent. No interest rate shall exceed a rate equal to the 20 bond index of tax exempt municipal bonds published by Credit Markets in New York, New York, on the Monday next preceding the day on which the bonds are sold, plus 2 percent. The difference between the highest rate specified and the lowest rate specified shall not exceed 2 percent. No bid of less than the par value of the bonds and accrued interest thereon to the date of delivery will be considered and no supplemental interest payments will be authorized. Each bid shall specify the total interest cost to the city during the life of the bond issue on the basis of such bid, the premium, if any, offered by the bidder, and the net interest cost to the city on the basis of such bid. Each bid shall also specify the average annual net interest rate to the city on the basis of such bid.

Basis of Award

The award of the bonds will be made on the basis of the lowest net interest cost to the city, which will be determined by subtracting the amount of the premium bid, if any, from the total interest cost to the city. If there is any discrepancy between the net interest cost and the average annual net interest rate specified, the specified net interest cost shall govern and the interest rates specified in the bid shall be adjusted accordingly. If two or more proper bids providing for identical amounts for the lowest net interest cost are received, the governing body shall determine which bid, if any, shall be accepted, and its determination shall be final.

Authority, Purpose and Security

The bonds are being issued pursuant to K.S.A. 12-6a01 et seq. and amendments thereto, inclusive, for the purpose of paying the cost of certain street improvements, alley repaying and sewer improvements. The bonds and the interest thereon will constitute general obligations of the city, payable as to both principal and interest in part from special assessments levied upon the property benefited by the improvements and, if not so paid, from ad valorem taxes which may be levied without limitation as to rate or amount upon all the taxable tangible property located within the city, the balance payable from ad valorem taxes which may be levied without limitation as to rate or amount upon all the taxable tangible property, real and personal, within the territorial limits of the city.

Legal Opinion and Tax Exemption

The bonds will be sold subject to the legal opinion of Gaar & Bell, Overland Park, Kansas, bond counsel, whose approving legal opinion as to the validity of the bonds will be furnished and paid for by the city, printed on the bonds and delivered to the successful bidder as and when the bonds are delivered.

The Internal Revenue Code of 1986 imposes certain requirements and restrictions which must be met subsequent to the issuance of state and local government obligations in order to maintan the exemption from federal income taxation of the interest on such obligations. In the opinion of Gaar & Bell, bond counsel, under existing statutes, regulations, rulings and judicial decisions, assuming continued compliance by the city, with the terms of the bond ordinance, the interest on the bonds is exempt from federal income taxation except as follows:

(a) For purposes of computing the corporate alternative minimum tax, the alternative minimum taxable income for any corporation for any taxable year beginning in 1987, 1988 or 1989 must be increased by 50 percent of the amount (if any) by which the adusted net book income of the corporation exceeds such corporation's alternative minimum taxable income for the taxable year (determined without regard to this adustment and alternative tax net operating loss deduction). For any taxable year beginning after 1989, the use of "adjusted net book income" will be replaced by "adjusted current earnings," and "50%" will be replaced by "75%." Interest on the bonds will be included in the calculation of adjusted net book income and adjusted current earnings.

(b) For taxable years beginning after December 31, 1986, property and casualty insurance companies will be required to reduce their deduction for losses incurred on insurance contracts by 15 percent of the amount of interest received or accrued on tax-exempt obligations acquired after August 7,

1986, including the bonds.

(c) H.R. 2005, the Superfund Amendments and Reauthorization Act of 1986, which was enacted on October 17, 1986, includes among its provisions the imposition of a new environmental tax on corporations. The amount of tax is equal to 0.12 percent of the excess of the modified alternative minimum taxable income of such corporation for the taxable year (determined without regard to the alternative tax net operating loss deduction and the deduction for the windfall profit tax allowed under Section 164(a)(5) of the code) over \$2 million. Interest on the bonds will be included in the calculation of the modified alternative minimum tax income of corporations. The environmental tax is imposed whether or not the taxpayer is subject to the alternative minimum tax, but is deductible from gross income. The environmental tax is effective for taxable years beginning after December 31, 1986 and before January 1, 1992. The imposition of this environmental tax could result in additional taxation of the interest on the bonds for certain bondowners.

Qualified Tax-Exempt Obligations

Under Section 265(b) of the code, for taxable years ending after December 31, 1986, financial institutions will not be allowed to deduct the portion of their interest expense which is "allocable" to tax-exempt interest on bonds acquired after August 7, 1986, unless

such bonds qualify for the exception described herein. The portion of interest expense allocable to tax-exempt interest is equal to the total interest expense of the taxpayer multiplied by the fraction:

Average adjusted bases (within the meaning of Section 1016 of the code) of the taxpayer's tax-exempt obligations acquired after August 7, 1986

Average adjusted bases for all of the taxpayer's assets

A bond which constitutes a "qualified tax-exempt obligation" is exempt from the operation of Section 265(b), and a financial institution may generally deduct 80 percent of its interest expense allocable to such a bond, as under prior law. A qualified tax-exempt obligation is a governmental purpose or qualified 501(c)(3) bond which is "designated" by the issuer for this purpose. No more than \$10,000,000 of bonds may be designated by the issuer for this purpose during any calendar year, and the reasonable anticipated amount of qualified tax-exempt obligations (other than private activity bonds) which will be issued by the issuer during the calendar year may not exceed \$10,000,000.

The city intends to designate the bonds as qualified tax-exempt obligations for purposes of Section 265 of the code.

Delivery and Payment

The city will pay for printing and registering the bonds and will deliver the same properly prepared, executed and registered without cost to the successful bidder by February 26, 1987 at such bank or trust company in the contiguous United States as may be specified by the successful bidder without cost to the successful bidder. Delivery elsewhere will be made at the expense of the purchaser. Said bidder will also be furnished with a certified transcript of the proceedings evidencing the authorization and issuance of the bonds and the usual closing proofs, which will include a certificate that there is no litigation pending or threatened at the time of delivery of the bonds affecting their validity. Payment for the bonds shall be made in federal reserve funds, immediately subject to use by the city.

Good Faith Deposit

Each bid shall be accompanied by a cashier's or certified check drawn on a bank located in the United States of America in the amount of \$21,300 payable to the order of the city to secure the city from any loss resulting from the failure of the bidder to comply with the terms of his bid. No interest will be paid upon the deposit made by the successful bidder. Said check shall be returned to the bidder if his bid is not accepted. If a bid is accepted, said check will be held by the city until the bidder shall have complied with all of the terms and conditions of this notice, at which time the check will be returned to the successful

bidder or paid to his order at the option of the city. If a bid is accepted but the city shall fail to deliver the bonds to the bidder in accordance with the terms and conditions of this notice, said check will be returned to the bidder. If a bid is accepted but the bidder shall default in the performance of any of the terms and conditions of this notice, the proceeds of such check will be retained by the city as and for liquidated damages.

CUSIP Numbers

It is anticipated that CUSIP identification numbers will be printed on certificated bonds or assigned to uncertificated bonds, but neither the failure to print such number on or assign such number to any bond nor any error with respect thereto shall constitute cause for failure or refusal by the purchaser thereof to accept delivery of and pay for the bonds in accordance with the terms of the purchase contract. All expenses in relation to the assignment and printing of CUSIP numbers on the bonds will be paid by the city.

Bond Rating

The outstanding general obligation bonds of the city are rated "A1" by Moody's Investor Service, Inc., and the city has applied for rating on the bonds herein offered for sale.

Bid Forms

All bids must be made on forms which may be procured from the city clerk. No additions or alterations in such forms shall be made and any erasures may cause rejection of any bid. The city reserves the right to waive irregularities and to reject any and all bids.

Submission of Bids

Bids must be submitted in sealed envelopes addressed to the undersigned city clerk and marked "Proposal for the Purchase of Internal Improvement Bonds." Bids may be submitted by mail or delivered in person to the undersigned at the City Hall and must be received by the undersigned prior to 10 a.m. C.S.T. January 27, 1987.

Official Statement

The city has prepared a preliminary official statement dated January 12, 1987, copies of which may be obtained from the city clerk. Upon the sale of the bonds, the city will adopt the final official statement dated February 1, 1987, and, at the request of the successful bidder, will furnish the successful bidder with a reasonable number of copies thereof without additional cost. Additional copies may be ordered by the successful bidder at his expense.

Assessed Valuation and Indebtedness

The total assessed valuation of the taxable tangible property within the city, for the year 1986, is \$142,185,079. The total general obligation indebtedness of the city as of the date of the bonds, including the bonds being sold, is \$23,672,000.

Additional Information

Additional information regarding the bonds may be obtained from the city clerk.

Dated January 12, 1987.

CITY OF HUTCHINSON, KANSAS
By Vernon Stallman, CMC
Director of Finance/City Clerk
City Hall
125 E. Avenue B
P.O. Box 1567
Hutchinson, KS 67504-1567
(316) 665-2614

Doc. No. 004979

(Published in the KANSAS REGISTER, January 22, 1987.)

NOTICE OF BOND SALE \$2,000,000 GENERAL OBLIGATION BONDS SERIES 1987 OF

UNIFIED SCHOOL DISTRICT 200 GREELEY COUNTY, KANSAS (TRIBUNE) (general obligation bonds payable

from unlimited ad valorem taxes)

Sealed Bids

Sealed bids will be received by the undersigned, Clerk of the Board of Education of Unified School District 200, Greeley County, Kansas, on behalf of the Board of Education at the district office, 400 W. Lawrence, Tribune, KS 67860, until 7:30 p.m. M.S.T. on Monday, February 2, 1987, for the purchase of \$2,000,000 principal amount of general obligation bonds, Series 1987, of the district hereinafter described. All bids will be publicly opened and read at said time and place and will be acted upon by the Board of Education. No oral or auction bids will be considered.

Bond Details

The bonds will consist of fully registered bonds in denominations of \$5,000 or any integral multiple thereof, dated February 1, 1987, and becoming due serially on October 1 in the years as follows:

Year	Principal Amount
1989	\$ 55,000
1990	60,000
1991	65,000
1992	70,000
1993	75,000
1994	75,000
1995	80,000
1996	85,000
1997	95,000
1998	100,000
1999	105,000
2000	110,000
2001	120,000
2002	130,000
2003	135,000
2004	145,000
2005	155,000

2006 165,000 2007 175,000

The bonds will bear interest from the date thereof at rates to be determined when the bonds are sold as hereinafter provided, which interest will be payable semiannually on April 1 and October 1 in each year, beginning on April 1, 1988.

Place of Payment and Bond Registration

The principal of and interest on the bonds will be payable in lawful money of the United States of America by check or draft of the Treasurer of the State of Kansas, Topeka, Kansas (the paying agent and bond registrar), to the registered owners thereof whose names are on the registration books of the bond registrar as of the 15th day of the month preceding each interest payment date. The bonds will be registered pursuant to a plan of registration approved by the district and the Attorney General of the State of Kansas.

The district will pay for the fees of the bond registrar for registration and transfer of the bonds and will also pay for printing a reasonable supply of registered bond blanks. Any additional costs or fees that might be incurred in the secondary market, other than fees of the bond registrar, will be the responsibility of the bondowners.

The number and denominations of the bonds and the names, addresses and social security or taxpayer identification numbers of the registered owners shall be submitted in writing by the successful bidder to the district and bond registrar at least two weeks prior to the closing date. In the absence of such information, the district will deliver one bond per maturity registered in the name of the manager of the successful bidder. The initial reoffering price to the public by the original purchaser shall be furnished to the district at least one week prior to the closing date. A certificate setting forth such initial reoffering price to the public shall be furnished to the district by the original purchaser at closing.

Redemption of Bonds Prior to Maturity

Bonds maturing in the years 1989 to 1995, inclusive, shall become due without option of prior payment. At the option of the district, bonds maturing in the years 1996 and thereafter may be called for redemption and payment prior to maturity in whole or in part in inverse order of maturity (selection of bonds within the same maturity to be by lot by the district in such equitable manner as it may determine) on October 1, 1995, or on any interest payment date thereafter at the redemption price of 101 percent (expressed as a percentage of the principal amount), plus accrued interest to the redemption date.

Whenever the district is to select the bonds for the purpose of redemption, it shall, in the case of bonds in denominations greater than \$5,000, if less than all of the bonds then outstanding are to be called for redemption, treat each \$5,000 of face value of each such fully registered bond as though it were a separate bond of the denomination of \$5,000.

If the district shall elect to call any bonds for re-

demption and payment prior to the maturity thereof, the district shall give written notice of its intention to call and pay said bonds on a specified date, said notice to be mailed by United States registered or certified mail addressed to the registered owners of said bonds, to the Treasurer of the State of Kansas, Topeka, Kansas, and to the manager or managers of the underwriting account making the successful bid, each of said notices to be mailed not less than 30 days prior to the date fixed for redemption. If any bond be called for redemption and payment as aforesaid, all interest on such bond shall cease from and after the date for which such call is made, provided funds are available for its payment at the price hereinbefore specified.

Conditions of Bids

Proposals will be received on the bonds bearing such rate or rates of interest as may be specified by the bidders, subject to the following conditions: The same rate shall apply to all bonds of the same maturity. Each interest rate specified shall be a multiple of 1/8 or 1/20 of 1 percent. No interest rate shall exceed a rate equal to the 20 bond index of tax exempt municipal bonds published by Credit Markets in New York, New York, on the Monday next preceding the day on which the bonds are sold, plus 2 percent. The difference between the highest rate specified and the lowest rate specified shall not exceed 2 percent. No bid of less than the entire par value of the bonds and accrued interest thereon to the date of delivery will be considered and no supplemental interest payments will be authorized. Each bid shall specify the total interest cost to the district during the life of the bonds on the basis of such bid, the premium, if any, offered by the bidder, and the net interest cost to the district on the basis of such bid, all certified by the bidder to be correct, and the district will be entitled to rely on the certificate of correctness of the bidder. Each bid shall also specify the average annual net interest rate to the district on the basis of such bid.

Basis of Award

The award of the bonds will be made on the basis of the lowest net interest cost to the district, which will be determined by subtracting the amount of the premium bid, if any, from the total interest cost to the district. If there is any discrepancy between the net interest cost and the average annual net interest rate specified, the specified net interest cost shall govern and the interest rates specified in the bid shall be adjusted accordingly. If two or more proper bids providing for identical amounts for the lowest net interest cost are received, the Board of Education shall determine which bid, if any, shall be accepted, and its determination shall be final. The district reserves the right to reject any and all bids and to waive any irregularities in a submitted bid.

Authority, Purpose and Security

The bonds are being issued pursuant to K.S.A. 72-6761 et seq., as amended, for the purpose of paying the cost of certain school building improvements. The bonds and the interest thereon will constitute general obligations of the district, payable from ad valorem taxes which may be levied without limitation as to rate

or amount upon all the taxable tangible property, real and personal, within the territorial limits of the district.

Legal Opinion and Tax Exemption

The bonds will be sold subject to the legal opinion of Gaar & Bell, Wichita, Kansas, bond counsel, whose approving legal opinion as to the validity of the bonds will be furnished and paid for by the district, printed on the bonds and delivered to the successful bidder as and when the bonds are delivered.

On October 22, 1986, the President of the United States signed into law H.R. 3838, the Tax Reform Act of 1986, which redesignates the Internal Revenue Code of 1954 as the Internal Revenue Code of 1986. The 1986 code imposes certain additional requirements and restrictions which must be met subsequent to the issuance of state and local government obligations in order to maintain the exemption from federal income taxation of the interest on such obligations. The district will covenant in the bond resolution to comply with the provisions of the Act and to take all action as may be necessary to comply with the Act and all applicable future law to preserve the tax-exempt status of the bonds, to the extent such actions can be taken by the Board of Education of the district.

In the opinion of Gaar & Bell, Wichita, Kansas, bond counsel, under existing law, statutes, regulations, rulings and judicial decisions, assuming continued compliance by the district with the terms of the bond resolution, the bonds are exempt from intangible personal property taxes levied by Kansas counties, cities or townships, and the interest on the bonds is exempt from federal income taxation except as follows:

(a) For taxable years beginning in the years 1987, 1988 and 1989, the interest on the bonds will be included in the adjusted net book income of corporations. For purposes of computing the corporate alternative minimum tax, a corporation's alternative minimum taxable income must be increased by 50 percent of the amount by which such corporation's adjusted net book income exceeds such corporation's alternative minimum taxable income (determined without regard to this adjustment or the alternative tax net operating loss deduction). For taxable years beginning after 1989, the use of "book income" will be replaced by "adjusted current earnings," and "50%" will be replaced by "75%."

(b) For taxable years beginning after December 31, 1986, property and casualty insurance companies will be required to reduce their deduction for losses incurred on insurance contracts by 15 percent of the amount of interest received or accrued on tax-exempt obligations acquired after August 7, 1986, including the bonds.

H.R. 2005, the Superfund Amendments and Reauthorization Act of 1986, which was enacted on October 17, 1986, includes among its provisions the imposition of a new environmental tax. Calculation of the tax is to be based generally on a percentage of the corporate alternative minimum taxable income as defined in the 1986 code which would include interest on tax-exempt obligations, including the bonds. The

amount of tax is equal to 0.12 percent of excess of the alternative minimum taxable income (without regard to net operating losses and the deduction for the environmental tax) over \$2 million. The environmental tax is imposed whether or not the taxpayer is subject to the alternative minimum tax, but is deductible from gross income. The environmental tax is effective for taxable years beginning after 1991. The imposition of this environmental tax could result in additional taxation of interest on the bonds for certain bondowners.

A form of bond counsel's opinion is contained in the official statement of the district with respect to the bonds.

Qualified Tax Exempt Obligations

The Tax Reform Act of 1986, H.R. 3838, was signed into law by the President of the United States on October 22, 1986. The Act provides that banks and thrift institutions would be unable to deduct any portion of the interest cost of purchasing or carrying tax-exempt obligations (with certain exceptions) if such interest costs are incurred in taxable years ending after December 31, 1986 with respect to bonds acquired after August 7, 1986. The Act provides that certain "qualified tax-exempt obligations" as defined in Section 902(b)(3) will be treated as having been acquired on August 7, 1986. The district will covenant to take such actions as are necessary to designate the bonds as "qualified tax-exempt obligations" described above.

Delivery and Payment

The district will pay for printing the bonds and will deliver the same properly prepared, executed and registered without cost to the successful bidder on or before February 27, 1987, at such bank or trust company in the state of Kansas or Kansas City, Missouri, as may be specified by the successful bidder. The successful bidder will also be furnished with a certified transcript of the proceedings evidencing the authorization and issuance of the bonds and the usual closing proofs, which will include a certificate that there is no litigation pending or threatened at the time of delivery of the bonds affecting their validity. Payment for the bonds shall be made in federal reserve funds, immediately subject to use by the district.

Good Faith Deposit

Each bid shall be accompanied by a cashier's or certified check drawn on a bank located in the United States of America equal to 2 percent of the total amount of the bid payable to the order of the district. If a bid is accepted, said check or the proceeds thereof will be held by the district until the bidder shall have complied with all of the terms and conditions of this notice. If a bid is accepted but the district shall fail to deliver the bonds to the bidder in accordance with the terms and conditions of this notice, said check or the proceeds thereof will be returned to the bidder. If a bid is accepted but the bidder defaults in the performance of any of the terms and conditions of this notice, the proceeds of such check will be retained by the district as and for liquidated damages. No interest

will be paid upon the successful bidder's good faith check.

CUSIP Numbers

It is anticipated that CUSIP identification numbers will be printed on certificated bonds or assigned to uncertificated bonds, but neither the failure to print such number on or assign such number to any bond nor any error with respect thereto shall constitute cause for failure or refusal by the purchaser thereof to accept delivery of and pay for the bonds in accordance with the terms of the purchase contract. All expenses in relation to the assignment and printing of CUSIP numbers on the bonds will be paid by the district.

Bid Forms

All bids must be made on forms which may be procured from the clerk or the financial adviser. No additions or alterations in such forms shall be made and any erasures may cause rejection of any bid. The district reserves the right to waive irregularities and to reject any and all bids.

Submission of Bids

Bids must be submitted in sealed envelopes addressed to the undersigned clerk and marked "Bond Bid." Bids may be submitted by mail or delivered in person to the undersigned at the district office and must be received by the undersigned prior to 7:30 p.m. M.S.T. on February 2, 1987.

Official Statement

The district has prepared an official statement dated January 15, 1987, copies of which may be obtained from the clerk or from the financial adviser. Upon the sale of the bonds, the district will furnish the successful bidder with a reasonable number of copies thereof without additional cost upon request. Additional copies may be ordered by the successful bidder at his expense.

Assessed Valuation and Indebtedness

The total assessed valuation of the taxable tangible property within the district, for the year 1986, is as follows:

Equalized assessed valuation of taxable,	
tangible property	\$35,188.135
Tangible valuation of motor vehicles	\$ 1,584,402
Equalized assessed tangible valuation for	
computation of bonded debt limitations	\$36,772,537

The total general obligation indebtedness of the district as of the date of the bonds, including the bonds being sold, is \$2,000,000.

Additional Information

Additional information regarding the bonds may be obtained from the clerk.

Dated January 5, 1987.

UNIFIED SCHOOL DISTRICT 200 GREELEY COUNTY, KANSAS (TRIBUNE) By Wanda L. Smith, Clerk 400 W. Lawrence Tribune, KS 67860 (316) 376-4211

(Published in the KANSAS REGISTER, January 22, 1987.)

HOUSE BILL No. 2049

An ACT concerning appropriations for the fiscal year ending June 30, 1987; lapsing certain amounts appropriated from the state general fund for certain state agencies; and imposing certain restrictions and limitations.

Be it enacted by the Legislature of the State of Kansas: Section 1.

ADJUTANT GENERAL

(a) On the effective date of this act, of the \$941,132 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the operational management account, the sum of \$25,316 is hereby lapsed.

(b) On the effective date of this act, of the \$1,360,144 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the physical plant opera-

tions account, the sum of \$73,075 is hereby lapsed.

(c) On the effective date of this act, of the \$162,640 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the emergency preparedness operations and administration account, the sum of \$7,500 is hereby lapsed.

On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the architect and engineering fees for construction of armories account, the sum of \$4,500 is hereby lapsed.

Sec. 2.

DEPARTMENT ON AGING

(a) On the effective date of this act, of the \$1,581,652 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the program grants account, the sum of \$522,408 is hereby lapsed.

Sec. 3.

STATE BOARD OF AGRICULTURE

(a) On the effective date of this act, of the \$5,191,555 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$124,374 is hereby lapsed.

(b) On the effective date of this act, of the \$1,498,829 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of

\$127,500 is hereby lapsed.

Sec. 4.

KANSAS ANIMAL HEALTH DEPARTMENT

(a) On the effective date of this act, of the \$183,320 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the animal disease control program account, the sum of \$9,900 is herebytlapsed.

Sec. 5.

COMMISSION ON CIVIL RIGHTS

(a) On the effective date of this act, of the \$609,230 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the promotion of civil rights account, the sum of \$28,112 is hereby lapsed.

Sec. 6.

ATTORNEY GENERAL

(a) On the effective date of this act, of the \$1,359,127 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$58,512 is hereby lapsed.

On the effective date of this act, of the \$254,667 appro-

priated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$5,675 is hereby lapsed.

(c) On the effective date of this act, of the \$650,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the additional operating expenditures for investigation and litigation regarding interstate water rights account, the sum of \$30,197 is hereby lapsed.

Sec. 7.

ATTORNEY GENERAL—KANSAS BUREAU OF INVESTIGATION

(a) On the effective date of this act, of the \$1,698,417 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$46,063 is hereby lapsed.

(b) On the effective date of this act, of the \$4,040,627 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$886 is hereby lapsed.

Sec 8

STATE CORRECTIONAL-VOCATIONAL. TRAINING CENTER

(a) On the effective date of this act, of the \$3,569,029 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$28,720 is hereby lapsed.

Sec. 9.

CORRECTIONS OMBUDSMAN BOARD

(a) On the effective date of this act, of the \$146,546 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the adult corrections oversight account, the sum of \$8,126 is hereby lapsed.

Sec. 10.

DEPARTMENT OF ADMINISTRATION

(a) On the effective date of this act, of the \$929,987 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the general administration account, the sum of \$48,135 is hereby lapsed.

(b) On the effective date of this act, of the \$4,111,305 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the accounting and reporting services account, the sum of \$142,449 is hereby lapsed.

(c) On the effective date of this act, of the \$937,222 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the budget analysis ac-

count, the sum of \$125,346 is hereby lapsed.

(d) On the effective date of this act, of the \$2,588,438 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the personnel services account, the sum of \$107,630 is hereby lapsed.

(e) On the effective date of this act, of the \$1,780,448 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the buildings and grounds

account, the sum of \$80,856 is hereby lapsed.

(f) On the effective date of this act, of the \$500 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the capitol area plaza authority account, the sum of \$250 is hereby lapsed.

(g) On the effective date of this act, of the \$3,092,800 appropriated for the above agency for the fiscal year ending June 30. 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the central management

85

information systems and computing operations account, the sum of \$134.993 is hereby lapsed.

(h) On the effective date of this act, of the \$1,279,433 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the purchasing and property management account, the sum of \$89,159 is hereby lapsed.

(i) On the effective date of this act, of the \$132,040 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the pooled money investment board account, the sum of \$3,759 is hereby lapsed.

(j) On the effective date of this act, of the \$1,521,566 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the architectural services and planning account, the sum of \$62,727 is hereby lapsed.

(k) On the effective date of this act, of the \$186,475 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the general services ac-

count, the sum of \$8,423 is hereby lapsed.

(l) On the effective date of this act, of the \$150,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the gubernatorial transition expenses account, the sum of \$5,700 is hereby lapsed.

(m) On the effective date of this act, of the \$250,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the executive aircraft operations account, the sum of \$664 is hereby lapsed.

(n) On the effective date of this act, of the \$25,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the planning for capital improvements—governor's residence account, the sum of \$6,250 is hereby lapsed.

Sec. 11.

STATE FIRE MARSHAL

(a) On the effective date of this act, of the \$1,199,064 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the fire safety and prevention account, the sum of \$48,819 is hereby lapsed.

Sec. 12.

FORT HAYS STATE UNIVERSITY

(a) On the effective date of this act, of the \$12,642,355 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$273,485 is hereby lapsed.

(b) On the effective date of this act, of the \$1,438,722 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of \$300,000 is hereby lapsed.

(c) On the effective date of this act, of the \$793,889 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student salaries and

wages account, the sum of \$26,004 is hereby lapsed.

(d) On the effective date of this act, of the \$33,700 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student off-campus work-study program account, the sum of \$10,000 is hereby lapsed.

Sec. 13.

KANSAS PUBLIC DISCLOSURE COMMISSION

(a) On the effective date of this act, of the \$182,657 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 17 of the 1986 Session Laws of

Kansas from the state general fund in the operating expenditures account, the sum of \$8,805 is hereby lapsed.

Sec. 14.

GOVERNOR'S DEPARTMENT

(a) On the effective date of this act, of the \$1,053,852 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the administration account, the sum of \$64,562 is hereby lapsed.

Sec. 15.

DEPARTMENT OF HEALTH AND ENVIRONMENT

(a) On the effective date of this act, of the \$12,979,358 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$520,096 is hereby lapsed.

(b) On the effective date of this act, of the \$400,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the prenatal care collaborative program account, the sum of \$15,200 is hereby lapsed.

(c) On the effective date of this act, of the \$1,354,506 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the aid to local units—general public health programs account, the sum of \$65,243 is

hereby lapsed.

(d) Notwithstanding the provisions of section 4(a) of chapter 29 of the 1986 Session Laws of Kansas providing for the fiscal year ending June 30, 1987, that not less than \$100,000 be expended from the aid to local units — general public health programs account of the state general fund for prenatal care collaborative programs, the above agency shall not expend more than \$34,757 from such account during such fiscal year for such programs.

Sec. 16.

KANSAS HIGHWAY PATROL

(a) On the effective date of this act, of the \$3,489,901 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 18 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$664,941 is hereby lapsed.

(b) On the effective date of this act, of the \$299,673 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 18 of the 1986 Session Laws of Kansas from the state general fund in the emergency medical services—state operations account, the sum of \$3,225 is hereby

lapsed.

(c) On the effective date of this act, of the \$120,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 18 of the 1986 Session Laws of Kansas from the state general fund in the aid to local units—EMS administrative grants account, the sum of \$6,000 is hereby lapsed.

(d) On the effective date of this act, the appropriation of \$80,000 made for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 18 of the 1986 Session Laws of Kansas from the state general fund in the expansion of EMS communications system account is hereby lapsed.

(e) On the effective date of this act, of the \$392,428 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 18 of the 1986 Session Laws of Kansas from the state general fund in the capitol area security account, the sum of \$12,500 is hereby lapsed.

Sec. 17.

STATE HISTORICAL SOCIETY

(a) On the effective date of this act, of the \$1,228,184 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the administration account, the sum of \$36,319 is hereby lapsed.

(b) On the effective date of this act, of the \$1,008,661 appro-

priated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the collections account, the sum of \$47,150 is hereby lapsed.

(c) On the effective date of this act, of the \$115,618 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the historic preservation

account, the sum of \$4,495 is hereby lapsed.

(d) On the effective date of this act, of the \$489,957 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the historic properties account, the sum of \$35,410 is hereby lapsed.

(e) On the effective date of this act, of the \$1,544,052 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the museum account, the

sum of \$78,521 is hereby lapsed.

(f) On the effective date of this act, of the \$100,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 28(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the historic properties maintenance and repairs account, the sum of \$24,000 is hereby lapsed.

Sec. 18.

DEPARTMENT OF HUMAN RESOURCES

(a) On the effective date of this act, of the \$1,016,998 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 41(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$47,514 is hereby lapsed.

(b) On the effective date of this act, of the \$212,477 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 41(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$1,967 is hereby lapsed.

Sec. 19.

DEPARTMENT OF COMMERCE

(a) On the effective date of this act, of the \$912,380 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the general administration (including official hospitality) account, the sum of \$30,944 is hereby lapsed.

(b) On the effective date of this act, of the \$226,750 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the grant for Kansas, Inc.

account, the sum of \$50,000 is hereby lapsed.

(c) On the effective date of this act, of the \$500,241 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the community development account, the sum of \$22,163 is hereby lapsed.

(d) On the effective date of this act, of the \$400,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the grants for certified development companies account, the sum of \$92,832 is hereby lapsed.

(e) On the effective date of this act, of the \$250,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the grants for small business development centers account, the sum of \$1,932 is hereby lapsed.

(f) On the effective date of this act, of the \$1,353,181 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the industrial development

account, the sum of \$28,088 is hereby lapsed.

(g) On the effective date of this act, of the \$807,329 appropriated for the above agency for the fiscal year ending June 30,

1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the travel, tourism and film services account, the sum of \$23,000 is hereby lapsed.

(h) On the effective date of this act, of the \$150,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the grants for small business innovative research account, the sum of \$50,000 is hereby lapsed.

(i) On the effective date of this act, of the \$610,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 30 of the 1986 Session Laws of Kansas from the state general fund in the Kansas advanced technology commission—research projects grants account, the sum of \$14,795 is hereby lapsed.

Sec. 20.

KANSAS CORRECTIONAL INSTITUTION

AT LANSING

(a) On the effective date of this act, of the \$2,914,514 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$8,300 is hereby lapsed.

(b) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the installation of emergency electrical generating equipment account, the sum of

\$38,924 is hereby lapsed.

(c) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the installation of fire sprinkler system in infirmary account, the sum of \$5,549 is hereby lapsed.

Sec. 21.

STATE INDUSTRIAL REFORMATORY

(a) On the effective date of this act, of the \$13,465,093 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$67,912 is hereby lapsed.

(b) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the capital improvement—plan and construct a 64-bed addition to the 96-bed minimum custody facility, including movable equipment account, the sum of \$82,000 is hereby lapsed.

Sec. 22.

YOUTH CENTER AT TOPEKA

(a) On the effective date of this act, of the \$6,055,988 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$39,666 is hereby lapsed.

Sec. 23.

YOUTH CENTER AT BELOIT

(a) On the effective date of this act, of the \$2,948,633 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$16,221 is hereby lapsed.

Sec. 24.

STATE BOARD OF INDIGENTS' DEFENSE SERVICES

(a) On the effective date of this act, of the \$2,981,550 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 19 of the 1986 Session Laws of Kansas from the state general fund in the indigents' defense services account, the sum of \$55,210 is hereby lapsed.

Sec. 25.

INSURANCE DEPARTMENT

(a) On the effective date of this act, of the \$3,599,155 appro-

priated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the insurance company regulation account, the sum of \$148,948 is hereby lapsed.

Sec. 26.

KANSAS COMMISSION ON INTERSTATE COOPERATION

(a) On the effective date of this act, of the \$106,561 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the national memberships and other operating expenditures (including official hospitality) account, the sum of \$5,477 is hereby lapsed.

Sec. 27.

JUDICIAL COUNCIL

(a) On the effective date of this act, of the \$151,985 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 19 of the 1986 Session Laws of Kansas from the state general fund in the judicial research account, the sum of \$13,824 is hereby lapsed.

Sec. 28.

YOUTH CENTER AT ATCHISON

(a) On the effective date of this act, of the \$3,590,669 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$20,343 is hereby lapsed.

Sec. 29.

KANSAS ARTS COMMISSION

(a) On the effective date of this act, of the \$585,750 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the promotion of the arts account, the sum of \$41,603 is hereby lapsed.

Sec. 30.

KANSAS NEUROLOGICAL INSTITUTE

(a) On the effective date of this act, of the \$8,837,522 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$299,358 is hereby lapsed.

Sec. 31.

KANSAS STATE UNIVERSITY

(a) On the effective date of this act, of the \$42,440,133 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$550,709 is hereby lapsed.

(b) On the effective date of this act, of the \$5,649,338 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of \$1,335,283 is hereby lapsed.

(c) On the effective date of this act, of the \$172,778 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the centers of excellence account, the sum of \$8,532 is hereby lapsed.

(d) On the effective date of this act, of the \$4,324,320 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures for utilities account, the sum of \$167,535 is hereby lapsed.

(e) On the effective date of this act, of the \$326,248 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the international grains program (including official hospitality) account, the sum of \$12,243 is hereby lapsed.

(f) On the effective date of this act, of the \$93,252 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of

Kansas from the state general fund in the international livestock program account, the sum of \$3,501 is hereby lapsed.

(g) On the effective date of this act, of the \$9,104,940 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the extension account, the sum of \$344,821 is hereby lapsed.

(h) On the effective date of this act, of the \$1,249,500 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the experimental fields

account, the sum of \$47,333 is hereby lapsed.

(i) On the effective date of this act, of the \$1,124,013 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student salaries and wages account, the sum of \$16,247 is hereby lapsed.

(j) On the effective date of this act, of the \$245,987 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the wheat research ac-

count, the sum of \$9,229 is hereby lapsed.

(k) On the effective date of this act, of the \$11,956,300 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the organized research account, the sum of \$451,036 is hereby lapsed.

(1) On the effective date of this act, of the \$896,474 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 55(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the southwest Kansas branch station operations account, the sum of \$34,062 is hereby lapsed.

(m) On the effective date of this act, of the \$468,167 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the Colby branch station operations account, the sum of \$17,773 is hereby lapsed.

(n) On the effective date of this act, of the \$1,050,721 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the Fort Hays branch station operations account, the sum of \$39,948 is hereby lapsed.

(o) On the effective date of this act, of the \$629,630 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the southeast Kansas branch station operations account, the sum of \$23,952 is hereby lapsed.

Sec. 32.

KANSAS STATE UNIVERSITY VETERINARY MEDICAL CENTER

(a) On the effective date of this act, of the \$838,201 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of \$195,413 is hereby lapsed.

(b) On the effective date of this act, of the \$843,042 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures for utilities account, the sum of \$34,166 is hereby lapsed.

Sec. 33.

EMPORIA STATE UNIVERSITY

(a) On the effective date of this act, of the \$12,831,767 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$206,265 is hereby lapsed.

(b) On the effective date of this act, of the \$1,858,919 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating ex-

penditures (including official hospitality) account, the sum of \$358,002 is hereby lapsed.

(c) On the effective date of this act, of the \$740,516 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures for utilities account, the sum of \$60,000 is hereby lapsed.

(d) On the effective date of this act, of the \$759,308 appropriated for the above agency for the fiscal year ending June 30. 1987, by section 5(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student salaries and wages account, the sum of \$2,485 is hereby lapsed.

Sec. 34.

PITTSBURG STATE UNIVERSITY

(a) On the effective date of this act, of the \$12,470,064 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$201,966 is hereby lapsed.

(b) On the effective date of this act, of the \$2,184,906 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of \$419,677 is hereby lapsed.

Sec. 35.

LARNED STATE HOSPITAL

(a) On the effective date of this act, of the \$18,835,496 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$241,955 is hereby lapsed.

Sec. 36.

LEGISLATIVE COORDINATING COUNCIL

(a) On the effective date of this act, of the \$208,111 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the legislative coordinating council-operations account, the sum of \$16,733 is hereby

On the effective date of this act, of the \$1,312,222 appro-(b) priated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the legislative research department—operations account, the sum of \$55,932 is hereby

lapsed.

(c) On the effective date of this act, of the \$1,394,986 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the office of revisor of statutes-operations account, the sum of \$66,021 is hereby lapsed.

(d) On the effective date of this act, of the \$50,108 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the legislative research department for the legislative educational planning committee-operations (including official hospitality) account, the sum of \$2,180 is hereby lapsed.

Sec. 37.

LEGISLATURE

(a) On the effective date of this act, of the \$6,331,764 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the operations (including official hospitality) account, the sum of \$259,116 is hereby lapsed.

Sec. 38.

STATE LIBRARY

(a) On the effective date of this act, of the \$139,387 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 22 of the 1986 Session Laws of

Kansas from the state general fund in the administrative services account, the sum of \$11,168 is hereby lapsed.

(b) On the effective date of this act, of the \$378,106 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the reference services account, the sum of \$10,460 is hereby lapsed

On the effective date of this act, of the \$593,155 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the Kansas library network services account, the sum of \$7,788 is hereby lapsed.

(d) On the effective date of this act, of the \$325,259 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the services to the blind

and physically handicapped account, the sum of \$450 is hereby lapsed.

(e) On the effective date of this act, of the \$127,451 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the interlibrary resource sharing and automation account, the sum of \$76,884 is hereby lapsed.

Sec. 39.

LIEUTENANT GOVERNOR

(a) On the effective date of this act, of the \$99,382 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 10(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the operations account, the sum of \$5,344 is hereby lapsed.

Sec. 40.

OSAWATOMIE STATE HOSPITAL

(a) On the effective date of this act, of the \$11,952,369 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$152,345 is hereby lapsed.

Sec. 41.

STATE PARK AND RESOURCES AUTHORITY

On the effective date of this act, of the \$25,060 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the administration account, the sum of \$25,060 is hereby lapsed.

(b) On the effective date of this act, of the \$1,949,231 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the state park operations

account, the sum of \$55,306 is hereby lapsed.

(c) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the new office building roof-Melvern state park-FY 1986 capital improvements account, the sum of \$650 is hereby lapsed.

(d) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 28 of the 1986 Session Laws of Kansas from the state general fund in the irrigation well-Lake Meade state park-FY 1985 capital improvements account, the

sum of \$1,250 is hereby lapsed.

Sec. 42.

PARSONS STATE HOSPITAL AND TRAINING CENTER

(a) On the effective date of this act, of the \$5,891,469 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$125,206 is hereby lapsed.

Sec. 43.

DEPARTMENT OF CORRECTIONS

(a) On the effective date of this act, of the \$1,619,062 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the central management

account, the sum of \$9,500 is hereby lapsed.

(b) On the effective date of this act, of the \$5,974,891 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the community services account, the sum of \$16,325 is hereby lapsed.

(c) On the effective date of this act, of the \$1,488,964 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the honor camps account,

the sum of \$19,100 is hereby lapsed.

(d) On the effective date of this act, of the \$4,475,468 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the community corrections account, the sum of \$727,247 is hereby lapsed.

(e) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the correctional capital improvements account, the sum of \$316,339 is hereby lapsed.

(f) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the capital improvements—plan and renovate third floor of Topeka prerelease center for 16 additional inmates account, the sum of \$46,000 is hereby lapsed.

(g) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the major repairs, special maintenance and remodeling for correctional institutions ac-

count, the sum of \$5,563 is hereby lapsed.

(h) On the effective date of this act, of the \$1,200,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 25(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the lease-purchase payment to the Ellsworth public building commission for the Ellsworth correctional work facility at Ellsworth, Kansas account, the sum of \$117,123 is hereby lapsed.

Sec. 44.

KANSAS PAROLE BOARD

(a) On the effective date of this act, of the \$332,186 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the parole from adult correctional institutions account, the sum of \$15,412 is hereby lapsed.

Sec. 45.

KANSAS STATE PENITENTIARY

(a) On the effective date of this act, of the \$20,451,619 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$275,360 is hereby lapsed.

(b) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the plan and construct medium-custody and support facilities, including remodeling of inside service building account, the sum of \$163,791 is hereby lapsed.

(c) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the installation of fire doors and alarms account, the sum of \$10,000 is hereby lapsed.

(d) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the replace locking system

in B cellhouse and renovate locking system in C cellhouse account, the sum of \$50,000 is hereby lapsed.

Sec. 46.

DIVISION OF POST AUDIT

(a) On the effective date of this act, of the \$1,088,758 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 20 of the 1986 Session Laws of Kansas from the state general fund in the operations (including legislative post audit committee) account, the sum of \$43,464 is hereby lapsed.

Sec. 47.

KANSAS PUBLIC BROADCASTING COMMISSION

(a) On the effective date of this act, of the \$2,500 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the operations account, the

sum of \$95 is hereby lapsed.

(b) On the effective date of this act, the appropriation of \$100,000 made for the above agency for the fiscal year ending June 30, 1987, by section 50(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the grant to Smoky Hills public television corporation for acquisition of KLBY television station in Colby, Kansas account is hereby lapsed.

(c) On the effective date of this act, the appropriation of \$639,500 made for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the public television—operating grants for existing stations account, the sum of \$15,498 is hereby lapsed.

Sec. 48.

STATE RECEPTION AND DIAGNOSTIC CENTER

(a) On the effective date of this act, of the \$3,192,348 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 10(a) of chapter 32 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$75,563 is hereby lapsed.

Sec. 49.

RAINBOW MENTAL HEALTH FACILITY

(a) On the effective date of this act, of the \$1,642,174 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$2,802 is hereby lapsed.

Sec. 50.

STATE BOARD OF REGENTS

(a) On the effective date of this act, of the \$683,581 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 10(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$29,960 is hereby lapsed.

(b) On the effective date of this act, of the \$167,200 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 10(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the contractual agreements—optometry education account, the sum of \$7,700 is

hereby lapsed.

(c) On the effective date of this act, of the \$4,625,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 10(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the scholarships and tuition grants account, the sum of \$197,000 is hereby lapsed.

Sec. 51.

STATE BOARD OF TAX APPEALS

(a) On the effective date of this act, of the \$637,786 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$30,675 is hereby lapsed.

Sec. 52.

DEPARTMENT OF REVENUE

(a) On the effective date of this act, of the \$15,318,380 appropriated for the above agency for the fiscal year ending June (continued)

30, 1987, by section 3(a) of chapter 16 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages

account, the sum of \$472,440 is hereby lapsed.

(b) On the effective date of this act, of the \$5,451,285 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 16 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$495,045 is hereby lapsed.

(c) On the effective date of this act, of the \$2,144,487 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 16 of the 1986 Session Laws of Kansas from the state general fund in the reappraisal operating expenditures account, the sum of \$156,478 is hereby lapsed.

(d) On the effective date of this act, of the \$8,000,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 16 of the 1986 Session Laws of Kansas from the state general fund in the aid to counties for reappraisal account, the sum of \$80,000 is hereby lapsed.

Sec. 53.

NORTON STATE HOSPITAL

(a) On the effective date of this act, of the \$3,037,024 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 10(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$94,735 is hereby lapsed.

Sec. 54.

KANSAS STATE SCHOOL FOR THE VISUALLY HANDICAPPED

(a) On the effective date of this act, of the \$2,024,028 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$24,425 is hereby lapsed.

Sec. 55.

KANSAS STATE SCHOOL FOR THE DEAF

(a) On the effective date of this act, of the \$3,962,832 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 5(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$158,992 is hereby lapsed.

Sec. 56.

SECRETARY OF STATE

(a) On the effective date of this act, of the \$425,372 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 12(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$51,848 is hereby lapsed.

(b) On the effective date of this act, of the \$815,059 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 12(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$1,439 is hereby lapsed.

Sec. 57.

DEPARTMENT OF SOCIAL AND REHABILITATION SERVICES

(a) On the effective date of this act, of the \$55,507,803 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the public assistance account, the sum of \$2,015,741 is hereby lapsed.

(b) On the effective date of this act, of the \$118,353,449 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the medical assistance

account, \$8,908,703 is hereby lapsed.

(c) On the effective date of this act, of the \$10,833,649 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the income maintenance account, the sum of \$66,321 is hereby lapsed.

(d) On the effective date of this act, of the \$2,575,727 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of

Kansas from the state general fund in the vocational rehabilitation account, the sum of \$44,705 is hereby lapsed.

(e) On the effective date of this act, of the \$617,276 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the blind services account, the sum of \$6,331 is hereby lapsed.

(f) On the effective date of this act, of the \$3,167,205 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the alcohol and drug abuse services account, the sum of \$12,910 is hereby lapsed.

(g) On the effective date of this act, of the \$18,570,070 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the administration

account, the sum of \$237,587 is hereby lapsed.

(h) On the effective date of this act, of the \$23,854,200 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the youth services account, the sum of \$255,826 is hereby lapsed.

(i) On the effective date of this act, of the \$5,332,771 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the adult services account.

the sum of \$74,730 is hereby lapsed.

(j) On the effective date of this act, of the \$1,055,317 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the mental health and retardation services—operating expenditures account, the sum of \$4,822 is hereby lapsed.

(k) On the effective date of this act, of the \$267,300 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the physician's professional liability insurance account, the sum of \$53,460 is hereby

lapsed.

(l) On the effective date of this act, of the \$47,574 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 3(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the student stipend programs for mental retardation institutions account, the sum of \$47,574 is hereby lapsed.

Sec. 58.

STATE CONSERVATION COMMISSION

(a) On the effective date of this act, of the \$201,975 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$2,493 is hereby lapsed.

(b) On the effective date of this act, of the \$1,300,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the water resources cost-sharing program account, the sum of \$129,399 is hereby lapsed.

(c) On the effective date of this act, of the \$1,000,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the for assistance in construction of watersheds account, the sum of \$1,684 is hereby lapsed.

(d) On the effective date of this act, of the \$200,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the state targeted cost-share program account, the sum of \$11,355 is hereby lapsed.

Sec. 59.

KANSAS SOLDIERS' HOME

(a) On the effective date of this act, of the \$1,081,331 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 13(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$79,203 is hereby lapsed.

Sec. 60.

DEPARTMENT OF EDUCATION

(a) On the effective date of this act, of the \$2,938,580 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$82,383 is hereby lapsed.

(b) On the effective date of this act, of the \$1,326,561 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$42,388 is hereby lapsed.

(c) On the effective date of this act, of the \$931,920 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the Fort Leavenworth school district account, the sum of \$33,264 is hereby lapsed.

(d) On the effective date of this act, of the \$451,656,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the state school equalization aid account, the sum of \$17,270,312 is hereby lapsed.

(e) On the effective date of this act, of the \$44,205,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the state school transportation aid account, the sum of \$1,686,936 is hereby lapsed.

(f) On the effective date of this act, of the \$20,766,495 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the community college credit hour state aid account, the sum of \$792,484 is hereby lapsed.

(g) On the effective date of this act, of the \$5,789,401 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the community college out-district state aid entitlement account, the sum of \$220,933 is hereby lapsed.

(h) On the effective date of this act, of the \$645,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the bilingual education programs aid account, the sum of \$24,614 is hereby lapsed.

(i) On the effective date of this act, of the \$7,132,963 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the area vocational-technical school program—state account, the sum of \$272,206 is hereby lapsed.

(j) On the effective date of this act, of the \$3,646,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the municipal university fund account, the sum of \$256,874 is hereby lapsed.

(k) On the effective date of this act, of the \$700,419 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the municipal university out-district state aid account, the sum of \$39,525 is hereby lapsed.

(1) On the effective date of this act, of the \$79,706,224 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the special education services aid account, the sum of \$3,033,096 is hereby lapsed.

(m) On the effective date of this act, of the \$85,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the educable deaf-blind and severely handicapped children's programs aid account, the sum of \$3,244 is hereby lapsed.

(n) On the effective date of this act, of the \$12,025,060 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws

of Kansas from the state general fund in the post-secondary aid for vocational education account, the sum of \$458,896 is hereby lapsed.

(o) On the effective date of this act, of the \$457,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the vocational education program aid—unified school districts account, the sum of \$67,440 is hereby lapsed.

(p) On the effective date of this act, of the \$25,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the Kansas foundation for agriculture project grant account, the sum of \$10,954 is hereby lapsed.

(q) On the effective date of this act, of the \$750,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the inservice education aid account, the sum of \$134,320 is hereby lapsed.

(r) On the effective date of this act, of the amount reappropriated for the above agency for the fiscal year ending June 30, 1987, by section 6(a) of chapter 22 of the 1986 Session Laws of Kansas from the state general fund in the municipal university state aid for student off-campus work-study program account, the sum of \$11,055 is hereby lapsed.

(s) On the effective date of this act, of the \$500,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 44(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the community college general state aid account, the sum of \$19,081 is hereby lapsed.

Sec. 61.

KANSAS TECHNICAL INSTITUTE

(a) On the effective date of this act, of the \$584,702 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of \$122,480 is hereby lapsed.

(b) On the effective date of this act, of the \$2,405,634 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$1,003 is hereby lapsed.

Sec. 62.

TOPEKA STATE HOSPITAL

(a) On the effective date of this act, of the \$11,400,186 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 11(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$351,915 is hereby lapsed.

Sec. 63.

STATE TREASURER

(a) On the effective date of this act, of the \$1,465,020 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 14(a) of chapter 17 of the 1986 Session Laws of Kansas from the state general fund in the administration and fiscal services account, the sum of \$60,777 is hereby lapsed.

Sec. 64.

JUDICIAL BRANCH

- (a) On the effective date of this act, of the \$4,117,349 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 19 of the 1986 Session Laws of Kansas from the state general fund in the administration of justice—appellate operations account, the sum of \$153,840 is hereby lapsed.
- (b) On the effective date of this act, of the \$36,286,887 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 19 of the 1986 Session Laws of Kansas from the state general fund in the administration of justice—district courts account, the sum of \$750,763 is hereby lapsed.

(c) On the effective date of this act, of the \$36,887 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 4(a) of chapter 19 of the 1986 Session Laws of Kansas from the state general fund in the judicial and professional review account, the sum of \$10,810 is hereby lapsed.

Sec. 65.

UNIVERSITY OF KANSAS

(a) On the effective date of this act, of the \$57,432,291 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$2,301,123 is hereby lapsed.

(b) On the effective date of this act, of the \$10,752,158 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of \$537,503 is hereby lapsed.

(c) On the effective date of this act, of the \$445 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the supplemental retirement benefits account, the sum of \$17 is hereby lapsed.

(d) On the effective date of this act, of the \$82,300 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 34(d) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the law school operating expenditures account, the sum of \$814 is hereby lapsed.

(e) On the effective date of this act, of the \$203,603 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the for renting Carruth O'Leary hall for office space account, the sum of \$7,841 is hereby lapsed.

(f) On the effective date of this act, of the \$176,483 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the capitol complex public management degree program account, the sum of \$4,085 is hereby lapsed.

(g) On the effective date of this act, of the \$1,193,879 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student salaries and wages account, the sum of \$35,160 is hereby lapsed.

(h) On the effective date of this act, of the \$154,210 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student off-campus work-study program account, the sum of \$28,000 is hereby lapsed.

(i) On the effective date of this act, of the \$207,073 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the fire service training program operations account, the sum of \$4,851 is hereby lapsed.

(j) On the effective date of this act, of the \$405,536 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the research program in tertiary oil recovery account, the sum of \$15,642 is hereby lapsed.

(k) On the effective date of this act, of the \$79,440 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the university affiliated research program at Parsons state hospital and training center account, the sum of \$3,239 is hereby lapsed.

(1) On the effective date of this act, of the \$3,700,192 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the geological survey account, the sum of \$142,955 is hereby lapsed.

(m) On the effective date of this act, of the \$205,918 appro-

priated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the equipment matching funds account, the sum of \$7,930 is hereby lapsed.

(n) On the effective date of this act, of the \$1,539,462 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the general research

account, the sum of \$72,615 is hereby lapsed.

(o) On the effective date of this act, of the \$173,128 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 7(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the centers of excellence account, the sum of \$6,670 is hereby lapsed.

Sec. 66.

UNIVERSITY OF KANSAS MEDICAL CENTER

(a) On the effective date of this act, of the \$34,326,151 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures (including official hospitality) account, the sum of \$1,417,181 is hereby lapsed.

(b) On the effective date of this act, of the \$647,708 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student salaries and

wages account, the sum of \$17,528 is hereby lapsed.

(c) On the effective date of this act, of the \$1,725,273 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the contractual agreements and scholarships for medical education account, the sum of \$251,490 is hereby lapsed.

(d) On the effective date of this act, of the \$9,488,946 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 9(a) of Chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the stipends, fringe benefits, contractual support and medical malpractice insurance premium expenditures for state-supported residents account, the sum of \$358,251 is hereby lapsed.

Sec. 67.

KANSAS COMMISSION ON VETERANS AFFAIRS

(a) On the effective date of this act, of the \$80,748 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 42(a) of chapter 33 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures account, the sum of \$15,189 is hereby lapsed.

Sec. 68

KANSAS WATER OFFICE

- (a) On the effective date of this act, of the \$792,031 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the salaries and wages account, the sum of \$1,627 is hereby lapsed.
- (b) On the effective date of this act, of the \$182,844 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 24 of the 1986 Session Laws of Kansas from the state general fund in the annual repayment on water storage costs on Clinton reservoir account, the sum of \$14,151 is hereby lapsed.

Sec. 69.

WINFIELD STATE HOSPITAL AND TRAINING CENTER

(a) On the effective date of this act, of the \$9,735,951 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 12(a) of chapter 27 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures account, the sum of \$89,860 is hereby lapsed.

Sec. 70.

WICHITA STATE UNIVERSITY

(a) On the effective date of this act, of the \$29,127,077 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 34 of the 1986 Session Laws

of Kansas from the state general fund in the salaries and wages

account, the sum of \$361,442 is hereby lapsed.

(b) On the effective date of this act, of the \$1,024,403 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the student salaries and wages account, the sum of \$89,214 is hereby lapsed.

(c) On the effective date of this act, of the \$2,954,796 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the other operating expenditures (including official hospitality) account, the sum of

\$840,492 is hereby lapsed.

(d) On the effective date of this act, of the \$2,667,297 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 8(a) of chapter 34 of the 1986 Session Laws of Kansas from the state general fund in the operating expenditures for utilities account, the sum of \$103,614 is hereby lapsed.

Sec. 71.

DEPARTMENT OF REVENUE—HOMESTEAD PROPERTY TAX REFUNDS

(a) On the effective date of this act, of the \$8,200,000 appropriated for the above agency for the fiscal year ending June 30, 1987, by section 2(a) of chapter 29 of the 1986 Session Laws of Kansas from the state general fund in the homestead tax refunds account, the sum of \$600,000 is hereby lapsed.

Sec. 72. If any provision or clause of this act or application thereof to any person or circumstances is held invalid, such invalidity shall not affect other provisions or applications of the act which can be given effect without the invalid provision or application, and to this end the provisions of this act are declared to be severable.

This act shall take effect and be in force from and after its publication in the Kansas register.

I hereby certify that the above BILL originated in the HOUSE, and passed that body January 14, 1987.

JAMES D. BRADEN Speaker of the House. GENEVA SEWARD Chief Clerk of the House.

Passed the SENATE January 15, 1987.

ROBERT V. TALKINGTON President of the Senate. LU KENNEY Secretary of the Senate.

APPROVED January 16, 1987.

MIKE HAYDEN Governor.

STATE OF KANSAS

Office of Secretary of State

I, BILL GRAVES, Secretary of State of the State of Kansas, do hereby certify that the above and foregoing is a correct copy of the original enrolled bill now on file in my office.

IN TESTIMONY WHEREOF, I have hereunto subscribed my name and affixed my official seal, this 16th day of January,

1987.

BILL GRAVES Secretary of State.

(SEAL)

KANSAS FACTS

Education is top priority

Education in Kansas springs from simple settingssod huts and one-room schoolhouses. If the means were modest, the goals were not. Early-day Kansans understood the value of education as revealed in the original State Constitution which provides: "The legislature shall encourage the promotion of intellectual, moral, scientific and agricultural improvement by establishing a uniform system of common schools, and schools of a higher grade, embracing normal, preparatory, collegiate and university departments.

From such humble beginnings, Kansas has developed a comprehensive network of public schools. The State Board of Education supervises all public elementary, secondary, vocational-technical schools and community colleges. The 10 members of the board, each representing different districts, are elected by the voters. They serve four-year terms, with five being elected every two years. Higher education in state universities is the responsibility of the State Board of Regents, a nine-member board appointed by the governor.

Elementary and Secondary Education

Elementary and secondary education is provided in Kansas primarily through the public school system. The state is divided into 306 unified school districts. Each district is controlled by a seven-member board of education, which is elected by the district residents. The unpaid board members serve four-year terms.

Local control of public educaton has a long history in Kansas. The state provides supervision, basic finance and related resources as well as general guidelines, but educational policies and operational matters are left to local school districts. Local boards are responsible for the quality of educational services available in their communities. They select their own textbooks and teaching materials, determine what is taught, establish graduation requirements and hire their own staff. The more than \$500 million spent annually on elementary and secondary schools is divided almost evenly between local property taxes and

The Kansas educational system was in the national spotlight on May 17, 1954 when the U.S. Supreme Court handed down its landmark decision in the case of Brown vs. Board of Education of Topeka. The decision held that "separate but equal" school facilities were inherently unequal and that school segregation violated the Constitution's guarantee under the 14th Amendment. Topeka had unusual segregation it existed only in grade schools. The city's junior highs and high schools had been integrated for several decades. Before the court even heard the case, complete integration of the Topeka schools was underway. The Brown decision set in motion the desegregation of schools throughout the United States.

Higher Education

Kansas ranks fifth in the United States in the number of institutions of higher learning per capita,

and eighth among the states in the number of public

institutions of higher learning per capita.

Bethel College, Newton, was the first Mennonite College in the nation. Haskell Indian Junior College in Lawrence is the only federally operated two-year college for Indians. Haskell celebrated its centennial in 1984. The school was dedicated September 17, 1884. Three days later, "the arrival of a wagon train bringing 42 Cheyennes and 36 Arapahoes created a stir at Haskell." The students came "with their parents wearing paint and feathers." Baker University, Baldwin City, is the oldest institution of higher learning in Kansas. Washburn University, Topeka, is the state's only public municipal university. It is home to one of the state's two law schools. The school's mascot is the "Ichabod," which is named for founder Ichabod Washburn.

Vocational-Technical Schools

Vocational-technical schools are the fastest growing educational area in the state. The 14 vocational-technical schools in Kansas are scattered throughout the state. They provide Kansas business and industry with skilled craftsmen and technicians. The state system covers seven general occupational areas: trade and industry, technical training, agriculture, distribution, home economics, business and office occupations and health occupations. Course offerings include agriculture production, distributive education, practical nursing, food management and production, clerical work, auto mechanics, carpentry, drafting, electronics, printing, welding and computer work.

Regents' Institutions

The seven Regents' institutions offer a variety of academic programs. Each institution has certain unique activities and offerings. A brief description of each follows:

University of Kansas, Lawrence—KU opened in 1866. It was the first state university in the Great Plains region. KU has an enrollment of nearly 27,000 students, the largest of any school in the Big Eight Conference. Its academic programs include the only medical school and the only pharmacy school in the state. It also has one of the two law schools in Kansas. The KU Medical Center, Kansas City, serves not only as a teaching facility for health professionals, but also as an important research and treatment hospital. KU's Mid-America Cancer Center is nationally recognized as one of the top cancer research and treatment centers, and a recently completed Radiation Therapy Center is one of the best in the country. The KU Medical Center operates a branch in Wichita. Special activities at KU include: the State Geological Survey, the Law Enforcement Training Center and the Bureau of Child Research, an internationally recognized leader in the study of mental retardation and developmental disabilities. KU athletic teams, the Jay-hawks, are members of the Big Eight Conference.

Kansas State University, Manhattan—Established in 1863, K-State was the nation's original land-grant university. Although the main campus is in Manhattan, the university owns or leases more than 14,000 acres in 19 counties for agricultural research. It also conducts agricultural programs in several foreign nations. More than 18,000 students are enrolled in the

university, which offers off-campus education in several areas of the state. K-State is especially noted for its academic programs in agriculture, home economics and veterinary medicine. Through the Federal Land-Grant Act, Kansas State is responsible for all county extension activities in the state. K-State has one of the most prestigious lecture series in American colleges and universities, the Alfred M. Landon Lecture Series on Public Issues. In athletic events, the K-State Wildcats compete in the Big Eight.

Wichita State University, Wichita—Founded as Fairmount College in 1895, WSU became a municipal university in 1926 and a state university in 1964. Enrollment is more than 17,000. WSU has a cultural program that serves not only the student body, but also the city of Wichita. Activities include theater, music, art exhibits and lectures, including the Dwight D. Eisenhower Lecture Series. The Center for Gerontology and the Center for Urban Studies provide special research and services. WSU's mascot is the Wheat Shocker. WSU is in the Missouri Valley Conference.

Emporia State University, Emporia—Established in 1863 as the Kansas State Normal School, ESU was later called the Kansas State Teachers College of Emporia. Training of educators is still emphasized, although other academic programs are offered. Enrollment is nearly 5,500. The ESU campus is designed to accommodate handicapped students, and an extensive program for the handicapped is available. ESU is also home to the Center for the Study of the Great Plains. The ESU Hornets are in the Central States Intercollegiate Conference.

Fort Hays State University, Hays—FHSU was established in 1901 as the western branch of the State Normal School. It is on land which was once the Fort Hays Military Reservation and is the only state university in the western two-thirds of the state. It has an enrollment of about 5,400. The FHSU Tigers are members of the Central States Intercollegiate Conference. Fort Hays burst into national basketball prominence in the 1980s. The Tigers made their first appearance at the NAIA national basketball tournament in Kansas City, Mo., in 1981 and in 1983 they finished third in the nation. In 1984 and 1985, they won backto-back national championships.

Pittsburg State University, Pittsburg—PSU was established in 1903 as the Auxiliary Manual Training Normal School, a branch of the Emporia Normal School. Enrollment is about 5,000. A unique part of the university is the School of Technology and Applied Science, which works toward improvement of various industries in Kansas. The school conducts seminars and workshops for industries and provides consultations on industrial safety. The Center for Business and Economic Development is another special program. PSU's athletic teams, the Gorillas, are members of the Central States Intercollegiate Conference.

Kansas Technical Institute, Salina—Opened in 1966, KTI offers two-year programs in solar energy technology, computer science technology and surveying technology, as well as more traditional fields of technical study. It has an enrollment of more than 600.

NOW AVAILABLE

CUSTOM-MADE LOOSELEAF BINDERS for the KANSAS REGISTER

KANG	×	
MANS		
	44.8	
	GIS	
State of W		
Vol. 1, No. 45	-43	
7A7		JACK
IN THIS ISSUE	11. 100	ocrotary BRIER
		77.000
		1191-1201
Notice to Bilders for State Purchase Kenne Fish and Game Commission Notice of Meeting Commission Attach	*****************	Age .
		Ligg
Attorney General's Opinions Opinions No. 85.934 dess		1193
Atterney Consent's Opinious Opinious De 254 through 22.23 Lagislative Interior Committee Schedule Notice of Sand Sand City of Systems City of Bells Plaine City of Opinious	***************************************	1183
City of Syracuse	*****	1195
City of Hutching		1195
City of Wint		1000
Supreme Court and Court of Appeals Dockets		1107
Appeals Dockets		1199
		190g

We are pleased to announce that custom-made Kansas Register binders are now available!

These binders will hold your copies of the Kansas Register attractively for permanent use. They are highest quality, durable, casebound Swing Hinge® binders made by McBee Loose Leaf Binder Products. (A Swing Hinge® binder has more capacity and allows for easier interfiling than standard ring binders.) They feature dark blue cloth covering and gold imprinting. Each three-inch binder will hold up to a year's worth of Register issues.

Order you	r binders today!							
Kansas Register binders \$14.00 each includes shipping and handling.								
••••••	CLIP AND MAIL	• • • • • • • • • • • • • • • • • • • •						
	Dear Secretary Brier: Please send K	Kansas Register binders.						
	Price: \$14.00 each, includes shipping an	nd handling.						
SHIP TO:	AMOUNT ENCLOSED \$							
		Shipping is by U.P.S. Delivery Service STREET ADDRESS IS NECESSARY.						
7								

Mail order, WITH PAYMENT, to: Kansas Register; Secretary of State; State Capitol; Topeka, KS 66612.

KANSAS REGISTER Secretary of State State Capitol Topeka, Kansas 66612-1594

Second Class postage paid at Topeka, Kansas

	Use this form (or a c SUBSCR			•	
	One-year subscriptions @ \$47.50 ea. TOTAL ENCLOSED			FOR REGISTER ONLY, PLEASE	OFFICE
	(Make checks payable to Kansas Register)		CODE	REC. NO.	
			EXPIRES	ENTERED BY	
END TO: Please, no nore than		· .			. 3.
address nes.)					
•	· ·				
	Zip code must be it	ncluded			
		. 4		•	
,	WITH PAYMENT, to: Kansas Register; Sec Use this form (or CHANGE OF	a copy of	of it) for SS		:
	Use this form (or	a copy of	of it) for SS	nitol; Topeka, KS	:
· · · · · · · · · · · · · · · · · · ·	Use this form (or CHANGE Of	a copy of	of it) for SS adicate change o		:
	Use this form (or CHANGE Of	a copy of	of it) for SS adicate change o		:
	Use this form (or CHANGE Of	a copy of	of it) for SS adicate change o		: